Hotelling Competition with Avoidable Horizontal Product Differentiation: Choosing Between Online Privacy and Disclosure

by

Shourjo Chakravorty* and David E. M. Sappington**

Abstract

We extend the standard model of Hotelling competition to allow consumers to change the horizontal characteristic of the product they purchase. Specifically, a consumer can incur a personal cost (K) to change the default privacy/disclosure setting on the smartphone she purchases. A unilateral increase in K can increase a supplier's equilibrium profit in some settings, despite rendering the supplier's phone less attractive to many potential customers. In other settings, an increase in K can reduce or have no effect on a supplier's equilibrium profit. Furthermore, a small increase in K can reduce a supplier's profit whereas a large increase in K can increase a supplier's profit.

Keywords: privacy choice; avoidable horizontal product differentiation.

JEL Codes: D43, D60, L13

June 2024

* Department of Economics, Faculty of Management, Istanbul Technical University Macka 34367, Istanbul, Türkiye (chakravorty@itu.edu.tr).

** Department of Economics, University of Florida PO Box 117140, Gainesville, Florida 32611 USA (sapping@ufl.edu).

1 Introduction.

There are presently more than 310 million smartphones in operation in the United States (Statista, 2023). 56.9% of these phones employ the iOS operating system. 42.6% employ the Android operating system (Statcounter, 2023). Apple installs the iOS operating system in the iPhone. Several device manufacturers, including Samsung, LG, HTC, and Sony, use the Android operating system in their smartphones. Both operating systems can collect data about users' online activities and/or allow other applications installed on user devices to collect these data. Relevant data include a user's real-time location, her information queries, the websites she visits, and the products she purchases online. Access to such information can help potential advertisers better assess the likely efficacy of targeted advertisements.

Some consumers are reluctant to share information about their online activities, in part because the information might be (mis)construed to reflect their personal beliefs and preferences.¹ Such consumers may prefer to share little, if any, information about their online activities. Other consumers may prefer to share information about some or all of their online activities with potential advertisers because doing so can increase the likelihood of receiving targeted, customized information that helps the consumers make informed purchasing decisions.²

Applications that run on an iPhone must obtain the user's permission to collect personal data that might be employed to inform targeted advertising.³ Thus, the default "privacy/disclosure" (PD) setting on the iPhone is "privacy," which means that personal data cannot be shared with advertisers without the user's explicit permission. In contrast, the default PD setting on many smartphones that employ the Android operating system (including the popular Samsung Galaxy phones) is "disclosure." This default PD setting implies that the user's personal data can be shared with advertisers unless the user explicitly prohibits such sharing.⁴

We employ a modified Hotelling model to analyze competition between two suppliers of smartphones with distinct default PD settings. In contrast to the standard model of Hotelling

¹Some consumers may prefer not to share information about their online activities so as to limit the number of unsolicited and undesired messages they receive from advertisers.

²See Tucker (2012), Taylor and Wagman (2014), and Acquisti et al. (2016), for example. Chellappa and Sin (2005) report that for many consumers, the perceived benefits from customized advertisements outweigh the perceived privacy costs from sharing personal information with advertisers. In contrast, Turow et al. (2009) find that these privacy costs exceed the associated benefits for a majority of consumers. Adlucent (2024), Salesforce Research (2024), and Statista (2024) present evidence regarding the value that consumers place on customized advertisements.

³https://developer.apple.com/app-store/user-privacy-and-data-use.

⁴See Grant and *Bloomberg* (2021) and Samsung (2023).

competition,⁵ we allow consumers to change the horizontal characteristic of the product they purchase. Specifically, by incurring personal cost K_1 , a consumer can change the default PD setting on Firm 1's phone from "privacy" to "disclosure." Similarly, by incurring personal cost K_2 , a consumer can change the default setting on Firm 2's phone from "disclosure" to "privacy." In practice, some consumers configure their communication devices to implement their stated preferences regarding privacy, but many consumers fail to do so.⁶ Such failure is consistent with the presence of substantial perceived personal costs of changing default PD settings on smartphones.

When a consumer can change the default PD setting on the phone she purchases, she can (at personal cost) effectively eliminate the horizontal product differentiation that she otherwise perceives. When default-switching costs are sufficiently low, the substantial horizontal product homogeneity that effectively prevails can fundamentally change the nature of the competition between suppliers. The firms can find it more profitable to focus on attracting all potential customers with a relatively low price than on attracting only "close" customers with a relatively high price.⁷

When relatively low default-switching costs induce a "market dominant" (MD) equilibrium in which all consumers purchase a phone from the same supplier, the effects of defaultswitching costs vary across suppliers. When Firm 1 attracts all consumers in equilibrium, its profit declines as its default-switching cost increases. The profit reduction arises in part because Firm 1 must reduce the price of its phone to continue to attract distant consumers who must now incur a higher cost to ensure their preferred PD setting. In contrast, when Firm 2 attracts all consumers in equilibrium, its profit can increase as its default-switching cost (K_2) increases. This is the case because when K_2 increases, fewer customers change Firm 2's default "disclosure" setting. Consequently, Firm 2 secures higher payments from advertisers, who pay a premium for access to the detailed personal information that is revealed under the "disclosure" setting.

Furthermore, for reasons that are explained in detail below, an increase in K_2 does not affect Firm 1's profit when Firm 1 serves all consumers in equilibrium. In contrast, when Firm 2 serves all consumers in equilibrium, its profit increases as K_1 increases. When both firms serve consumers in equilibrium, neither firm's profit is affected by changes in its own default-switching cost or in the rival's corresponding cost.

⁵For selective views of the literature on Hotelling competition, see Graitson (1982), Gabszewicz and Thisse (1992), and Biscaia and Mota (2013), for example.

⁶See Acquisti and Grossklags (2005), Athey et al. (2007), and Hutton and Ellis (2023), for example.

⁷ "Close" consumers are those with a relatively strong affinity for the firm's default PD setting. "Distant" consumers are those with a relatively strong affinity for the rival's default PD setting.

Even though a higher default-switching cost reduces the attraction of a firm's phone to some consumers, a unilateral increase in a firm's default-switching cost can enhance the firm's equilibrium profit. This is the case, for example, when the increased cost alters the nature of the prevailing equilibrium. As noted above, a higher default-switching cost can encourage a firm to focus on attracting (only) relatively close customers and charging them a relatively high price.⁸ Because prices are strategic complements under Hotelling competition, the ensuing relaxed price competition can enhance the profits of both industry suppliers in the "market sharing" (MS) equilibrium that prevails.⁹ We find that a higher default-switching cost is particularly likely to increase a firm's equilibrium profit when the firm's competitive advantage is limited and consumer transportation costs are relatively large (so the horizontal product characteristic is relatively important to consumers).¹⁰

MD and MS equilibria can both arise, regardless of whether default switching costs $(K_1$ and K_2) are endogenous or exogenous. However, certain equilibria that exist when K_1 and K_2 are exogenous do not arise when K_1 and K_2 are endogenous. To illustrate, there are conditions under which Firm 1 will not set $K_1 > 0$ in the MD equilibrium where it serves all consumers because, by setting $K_1 = 0$, Firm 1 could both increase the payments it receives from advertisers (by inducing expanded switching of Firm 1's default "privacy" setting) and allow Firm 1 to attract all consumers with a higher price (by reducing the cost that distant customers must incur to secure their preferred PD setting). Furthermore, there are conditions under which Firm 2 will not implement small values of K_2 because a higher default-switching cost would increase Firm 2's payment from advertisers by more than it would dissuade distant customers from purchasing Firm 2's phone.

This research contributes to the literature on Hotelling competition primarily by characterizing the changes that arise when consumers can incur a personal cost to eliminate the horizontal product differentiation that they otherwise perceive. In part because of its relevance and practical importance in today's "information age," the particular horizontal product differentiation that we consider pertains to the default PD setting on smartphones.¹¹

⁸A higher default-switching cost can also increase Firm 2's profit by inducing more customers to retain Firm 2's default "disclosure" setting, which ensures that Firm 2 secures higher payments from advertisers.

⁹Because a firm (Firm *i*) can benefit from an increase in its default-switching cost (K_i) that alters the nature of the prevailing equilibrium, a substantial increase in K_i can increase Firm *i*'s equilibrium profit even though a marginal increase in K_i can reduce its equilibrium profit.

¹⁰The UK Competition and Markets Authority (CMA) and the Australian Competition and Consumer Commission (ACCC) observe that platforms (including smartphone operating systems) can design their systems to increase the time and effort that consumers must expend to change platform default settings (CMA, 2020; ACCC, 2021). Our analysis can help to understand when such design enhances platform profit.

¹¹Rhodes and Zhou (2022) consider endogenous data sharing that can allow suppliers to practice perfect

However, our analysis and findings are relevant more generally.¹² Because our model differs from its predecessors primarily by incorporating default-switching costs, we emphasize how changes in these costs affect both the nature and the details of equilibrium outcomes. As noted above, we show that an increase in a firm's default-switching cost can increase, reduce, or have no effect on the firm's equilibrium profit.¹³

In our model, an increase in a firm's default-switching cost reduces the attraction of the firm's product to distant consumers by a fixed amount. Other authors (e.g., von Ungern-Sternberg, 1988; Hendel and de Figueiredo, 1997; Troncoso-Valverde and Robert, 2004; Hou et al., 2013) have examined the impact of an increase in consumer transportation costs, which reduces the attraction of a firm's product to all consumers. In these models, the extent of the reduced attraction increases linearly with the distance between a consumer's location and the relevant firm's location on the Hotelling line. These differences generate different conclusions about the conditions under which higher costs increase profits. However, in these models and in our model, this profit-enhancing effect of higher costs arises because the increased cost induces a firm to increase its focus on attracting particularly close customers.¹⁴

We develop and further explain our findings as follows. Section 2 describes our model. Section 3 characterizes equilibrium outcomes in two benchmark settings: one where default PD settings cannot be changed and one where consumers can change the default settings costlessly. Section 4 characterizes the distinct equilibria that can arise in the setting of primary interest, where default-switching costs are intermediate in magnitude. Section 5 examines the impacts of default-switching costs on equilibrium profits. Section 6 identifies conditions under which an increase in one supplier's default-switching cost increases the equilibrium

price discrimination. The authors show that the welfare effects of data sharing can differ with the extent of equilibrium market coverage. They also show that data sharing in excess of the welfare-maximizing level can arise in equilibrium.

¹²To illustrate, privacy considerations analogous to those that arise in the context of iPhones vs. Samsung Galaxy phones arise in the context of iPads and Samsung Galaxy tablets. Similar considerations also arise in the context of MacBooks/Macs and laptop/desktop computers that run on Microsoft Windows. Many other settings also arise in which a consumer can incur a cost to modify a default horizontal product characteristic. For example, a consumer might pay a detailing company either to change the color of an automobile that she has purchased, or to add or remove pinstripes from the car.

¹³Furthermore, an increase in one firm's default-switching cost can either increase or have no effect on the rival's equilibrium profit. Wilson (2010) identifies conditions under which a firm can enhance its profit by unilaterally increasing the time it takes for consumers to learn the firm's price. The resulting increased search costs for relatively "impatient" consumers induce them to learn the rival's price first and renders them less likely to subsequently learn the firm's price, thereby softening competition for more patient consumers. The softened competition in Wilson's model gives rise to conditions under which, as in our model, a firm can enhance its profit by unilaterally suppressing the demand for its product by a subgroup of consumers.

¹⁴Additional similarities and differences between our analysis and related analyses in the literature are discussed in Section 2, where we explain the details of our model.

profit of both suppliers. Section 7 examines equilibrium outcomes when default-switching costs are endogenous. Section 8 summarizes our key findings and suggests directions for future research. The Appendix provides the proofs of all formal conclusions in the text.

2 The Model

We analyze competition between two sellers of smartphones. A phone can operate under either a "privacy" setting or a "disclosure" setting. When a consumer uses a phone to access the internet under the "privacy" setting, the consumer's online activities (e.g., the sites she visits) are not revealed to potential advertisers. In contrast, the consumer's online activities are revealed to advertisers when her phone operates under the "disclosure" setting.

For the reasons discussed above, consumers' preferences for "privacy" and "disclosure" differ. These preferences are captured by a consumer's location on the unit interval. The consumer located at 0 has the strongest preference for privacy. The further is a consumer from 0, the more amenable is the consumer to disclosure. The consumer located at 1 has the strongest preference for disclosure. Potential consumers are distributed uniformly on the unit interval. The total mass of consumers is normalized to 1.

In standard Hotelling fashion, the two suppliers of smartphones are located at opposite ends of the unit interval: Firm 1 is located at 0, Firm 2 is located at 1. The default "privacy/disclosure" (PD) setting on Firm 1's phone is "privacy." The default PD setting on Firm 2's phone is "disclosure."¹⁵ If a consumer located at $x \in [0,1]$ purchases a phone from Firm 1 (respectively, Firm 2) and does not change the default PD setting on the phone she purchases, the consumer incurs "transportation" cost tx (respectively, t[1-x]). t > 0is a parameter that reflects the diversity and intensity of consumers' preferences for privacy vs. disclosure.¹⁶

In contrast to the standard Hotelling analysis, we allow consumers to change the default PD setting on the phones they purchase. $K_i \ge 0$ is the personal cost that a customer must incur to change the default PD setting on a phone she purchases from Firm $i \in \{1, 2\}$. This cost might reflect, for example, the time and effort required to learn how to change the phone's default setting, and then implement the change.¹⁷

¹⁵We take the firms' distinct PD choices to be exogenous. See Gal-Or et al. (2018), for example, for an analysis of how competing platforms choose the levels of privacy that they provide to their users.

¹⁶ "Transportation" costs associated with departures from preferred PD settings can be significant in practice. The CMA finds that many consumers cite "privacy and security" as a major determinant of the choice of a smartphone (CMA, 2022). Beyond Identity (2022) reports that many Android users who consider switching to an iPhone do so primarily because of the iPhone's perceived superiority in security and privacy.

¹⁷In practice, firms can influence the costs that consumers incur to change the default PD setting in part by choosing the nature and the number of steps that must be undertaken to implement the change. See, for

The default PD setting on a phone reflects a horizontal product characteristic. The phones that Firms 1 and 2 sell also can differ on vertical characteristics (e.g., processor speed, memory, battery life, photographic capabilities, and ease of use). G_i is the gross value that each customer derives from a phone supplied by Firm $i \in \{1, 2\}$, a value that primarily reflects the phone's vertical characteristics. The utility that a consumer derives from a phone purchased from Firm i is G_i , less any transportation and default-switching costs the consumer incurs, less p_i , which is the price that Firm i charges for its phone.

Firm *i* incurs unit production cost c_i . In addition to the revenue it collects from customers, each firm receives payments from advertisers. Firm *i* receives payment $r_H > 0$ (respectively, $r_L \ge 0$) from advertisers for each phone it sells when the purchaser employs the phone under the disclosure (respectively, the privacy) PD setting. Following Casadesus-Masanell and Hervas-Drane (2015), we take $\Delta \equiv r_H - r_L$ to be strictly positive. This assumption reflects advertisers' willingness to pay a premium for detailed information about a consumer's preferences, as revealed by her online activities.¹⁸

Each customer buys at most one phone. We further assume that G_1 and G_2 are sufficiently large relative to c_1 , c_2 , and t that every potential consumer purchases a phone in equilibrium. We define $A \equiv \frac{1}{3} [G_2 - c_2 + r_H - (G_1 - c_1 + r_L)]$ and refer to Firm 2 as the *advantaged firm* (with *competitive advantage* A) when A > 0, whereas Firm 1 is the advantaged firm (with competitive advantage |A|) when A < 0.

The timing in the model is as follows. After G_i , c_i , K_i $(i \in \{1, 2\})$, t, r_L , r_H , and default PD settings are determined exogenously,¹⁹ the two suppliers set their prices simultaneously and noncooperatively. Consumers then decide whether to purchase a phone from Firm 1 or from Firm 2. Next, each customer decides whether to retain or change the default PD setting on the phone she has purchased. Finally, the firms collect the stipulated payments from advertisers.²⁰

Each consumer purchases the phone that ensures her the highest utility (gross value, less

example, Athey et al. (2017), CMA (2020), and ACCC (2021).

¹⁸See Beales (2010), Mueller and Castro (2021), and Arora et al. (2021), for example. In practice, an online advertiser typically makes a payment to a platform only when a consumer on the platform clicks on the advertiser's message (Danao and Main, 2024). Consumers tend to be far more likely to click on a targeted (customized) message than to click on an untargeted message (Yan et al., 2009; Farahat and Bailey, 2012). Consequently, platforms typically receive more revenue from advertisers that deliver targeted messages than from advertisers that deliver untargeted messages.

¹⁹Section 7 allows K_1 and K_2 to be endogenous.

²⁰Because our model is static and each consumer purchases at most one phone, we do not consider how a firm might "poach" the customers of a rival supplier or set different prices for "new" and "existing" customers in a setting with customer switching costs (e.g., Gehrig and Stenbacka, 2004; Arie and Grieco, 2014; Rhodes, 2014).

price, less relevant transportation and default-switching costs). The consumer located at $x \in [0, 1]$ purchases a phone from Firm 1 rather than from Firm 2 if:

$$G_1 - p_1 - \min\{tx, t[1-x] + K_1\} > G_2 - p_2 - \min\{t[1-x], tx + K_2\}.$$
 (1)

The consumer located at $x \in [0, 1]$ purchases a phone from Firm 2 rather than from Firm 1 if the inequality in (1) is reversed.²¹ (1) reflects the fact that, after purchasing a phone from Firm 1, the consumer located at x will change the default PD setting on the phone ("privacy") if and only if the sum of the default-switching cost on Firm 1's phone and the consumer's transportation cost associated with the "disclosure" setting is less than the consumer's transportation associated with the "privacy" setting, i.e.:

$$t \left[1 - x \right] + K_1 < t x \quad \Leftrightarrow \quad x > \frac{1}{2} + \frac{K_1}{2t}.$$

$$\tag{2}$$

(2) implies that if $K_1 < t$, the (only) consumers who will change the default PD setting on a phone they purchase from Firm 1 are those located closest to Firm 2, i.e., those located in $(\frac{1}{2} + \frac{K_1}{2t}, 1].$

Similarly, after purchasing a phone from Firm 2, the consumer located at x will change the default PD setting on the phone ("disclosure") if and only if:²²

$$tx + K_2 < t[1-x] \Leftrightarrow x < \frac{1}{2} - \frac{K_2}{2t}.$$
(3)

Most of the ensuing analysis will consider settings in which $K_i \in (0,t)$ for $i \in \{1,2\}$. These settings avoid the relatively uninteresting case in which default-switching costs exceed the transportation cost associated with traversing the entire unit interval. In this case, no consumer would ever change the default PD setting on a phone she purchased.

For expositional ease, we restrict attention to undominated strategies for Firms 1 and 2. Formally, we assume that no firm ever sets a price that would never allow it to secure nonnegative profit, regardless of the price set by the rival supplier.

3 Benchmark Settings

In this section, we briefly characterize equilibrium outcomes in two benchmark settings. The first benchmark reflects the standard Hotelling model in which all product characteristics (including the default PD setting) are immutable. The second benchmark reflects the other extreme in which each consumer can costlessly change the default PD setting on the phone

²¹We assume that when a consumer is indifferent between purchasing a phone from Firm 1 and from Firm 2, she purchases a phone from each firm with probability $\frac{1}{2}$.

²²For expositional ease, we assume that when a consumer is indifferent between retaining and changing the default PD setting on the phone she purchases, she retains the default setting.

she purchases.

Lemma 1 characterizes outcomes in the setting where the default PD setting cannot be changed. The lemma refers to π_i , which denotes the profit of Firm $i \in \{1, 2\}$.

Lemma 1. Suppose the default PD setting cannot be changed and t > |A|. Then in equilibrium: (i) the consumer located at $x_0 \equiv \frac{1}{2} - \frac{A}{2t} \in (0,1)$ is indifferent between buying a phone from Firm 1 and from Firm 2; (ii) all consumers located in $[0, x_0)$ buy a phone from Firm 1; and (iii) all consumers located in $(x_0, 1]$ buy a phone from Firm 2. Furthermore: $p_1 = c_1 - r_L + t - A$; $p_2 = c_2 - r_H + t + A$; $\pi_1 = \frac{1}{2t} [t - A]^2$; and $\pi_2 = \frac{1}{2t} [t + A]^2$.

Lemma 1 considers settings in which consumers' preferences for privacy vs. disclosure are relatively pronounced in the sense that the unit transportation cost exceeds each firm's competitive advantage (i.e., t > |A|). When all product characteristics are immutable in this setting, those consumers with the strongest preference for privacy (respectively, disclosure) purchase a phone from Firm 1 (respectively, Firm 2). The advantaged firm (i.e., Firm 2 when A > 0 and Firm 1 when A < 0) sells more phones and secures greater profit than its rival.²³ Furthermore, equilibrium prices increase with own production costs $(\frac{\partial p_i}{\partial c_i} > 0$ for $i \in \{1,2\}$), with transportation costs $(\frac{\partial p_i}{\partial t} > 0)$, and with a firm's competitive advantage $(\frac{\partial p_1}{\partial |A|} > 0$ when A < 0 and $\frac{\partial p_2}{\partial A} > 0$ when A > 0). Furthermore, prices decline as payments from advertisers increase $(\frac{\partial p_1}{\partial r_L} < 0$ and $\frac{\partial p_2}{\partial r_H} < 0$) because these payments enhance each firm's incentive to expand its sales by reducing its price.

Lemma 2. Suppose $K_1 = K_2 = 0$. Then in equilibrium, all consumers purchase a phone from Firm 1 if $G_1 - c_1 > G_2 - c_2$. In contrast, all consumers purchase a phone from Firm 2 if $G_2 - c_2 > G_1 - c_1$. Consumers located in $[0, \frac{1}{2})$ implement the privacy setting on the phone they purchase, whereas consumers located in $(\frac{1}{2}, 1]$ implement the disclosure setting.²⁴ When all consumers purchase a phone from Firm *i*, the firm's profit is (nearly) $G_i - c_i - (G_j - c_j)$ for $i, j \in \{1, 2\}$ $(j \neq i)$.

When consumers can costlessly change the default PD setting on the phone they purchase, they effectively perceive the two phones to have the same horizontal product characteristic.

²³This outcome reflects the relatively high gross value that consumers derive from the advantaged firm's phone, the advantaged firm's relatively low production cost, and/or the relatively high advertising revenue associated with the firm's default PD setting.

²⁴The consumer located at $\frac{1}{2}$ is indifferent between the two PD settings. Consequently, by assumption, this consumer retains the default PD setting on the phone she purchases.

Each consumer simply implements her preferred PD setting on the phone she purchases, so the two phones are equally effective at satisfying each consumer's PD preference.²⁵ The effective absence of horizontal product differentiation leads to intense "winner-take-all" price competition. When $G_1 - c_1 > G_2 - c_2$, for example, Firm 1 reduces its price to the highest level that allows Firm 1 to attract all consumers when Firm 2 sets the lowest price at which it could profitably serve all consumers (i.e., $p_2 = c_2 - \frac{1}{2} [r_L + r_H]$).²⁶ (1) implies that this price for Firm 1 is determined by $G_1 - p_1 = G_2 - p_2$. Therefore, $p_1 = G_1 - G_2 + c_2 - \frac{1}{2} [r_L + r_H]$, which generates profit $p_1 - c_1 + \frac{1}{2} [r_L + r_H] = G_1 - c_1 - (G_2 - c_2)$ for Firm 1.

When $K_1 = K_2 = 0$, equilibrium prices and profits do not vary with payments from advertisers. This is the case because when each firm effectively competes to serve all consumers and when every consumer implements her preferred PD setting, each firm anticipates the same total payment from advertisers $(\frac{1}{2}[r_L + r_H])$. The relatively intense competition between suppliers of a product with no horizontal product differentiation effectively compels the firms to pass along all payments from advertisers to consumers.

4 Characterizing Equilibrium Outcomes

We now characterize equilibrium outcomes in the setting of primary interest where default-switching costs are intermediate in magnitude. Specifically, $K_1 \in (0, t)$ and $K_2 \in (0, t)$, so: (i) it is costly for consumers to change the default PD setting on the phone they purchase; and (ii) default-switching costs are less than the transportation cost of traversing the entire unit interval.

Proposition 1 identifies conditions under which a *market-sharing equilibrium* (i.e., an equilibrium in which both firms sell a strictly positive number of phones) arises. These conditions include the following:

Condition 1A. $t \ge \max\{r_H - c_2 - A, r_L - c_1 + A\}.$

Condition 1B. $\frac{1}{2t} [t-A]^2 > -2A + \frac{t-K_1}{2t} [2t+r_H-r_L]$ if $A \le 0$; $\frac{1}{2t} [t+A]^2 > 2A + \frac{t-K_2}{2t} [2t-(r_H-r_L)]$ if $A \ge 0$.

Condition 1A helps to ensure that prices are positive in the market-sharing equilibrium.

 $^{^{25}}$ For consumers located in (0, 1), neither phone delivers the consumer's ideal PD level. However, when it is costless to change each phone's default PD setting, every consumer perceives the two phones to be equally effective at meeting her PD preference.

 $^{{}^{26}\}frac{1}{2}[r_L + r_H]$ is the payment from advertisers that a firm receives when it serves all consumers and each consumer implements her preferred PD setting.

Condition 1B ensures that each firm earns more profit in the market-sharing equilibrium identified in Proposition 1 than it could secure by unilaterally lowering its price to the level required to secure the patronage of all consumers.²⁷

Proposition 1. Suppose: (i) t > |A|; (ii) $K_2 \in (A, t)$ if A > 0; and (iii) $K_1 \in (|A|, t)$ if A < 0. Further suppose that Conditions 1A and 1B hold. Then an equilibrium exists in which the outcomes identified in Lemma 1 prevail and no consumer changes the default setting on the phone she purchases.

Proposition 1 indicates that once the default-switching cost of the advantaged firm is sufficiently pronounced, the firm finds it more profitable to set a relatively high price and only attract "close" consumers than to set the lower price required to attract all consumers.²⁸ It typically becomes less profitable for a firm to attract distant consumers as default-switching costs increases. This is the case because the firm must reduce its price to convince distant consummers to either accept their less-preferred PD setting or incur the higher default-switching cost required to implement their preferred PD setting. Once its default-switching cost increases above a critical level, the advantaged firm effectively cedes distant consumers to the rival firm by setting a relatively high price that is attractive only to close consumers. The close consumers that the advantaged firm serves in equilibrium are either: (i) those that prefer the firm's default PD setting; or (ii) those whose preference for the rival's default PD setting is sufficiently mild that they choose not to change the default PD setting on the phone they purchase. Consequently, no consumer changes the default PD setting on the phone she purchases, even though such default switching is not prohibitively costly (i.e., even though $K_1 < t$ and $K_2 < t$). Therefore, prices, outputs, and profits do not vary with default-switching costs in this equilibrium.

In contrast, equilibrium profits do vary with default-switching costs in *market-dominant* equilibria, which are equilibria in which all consumers purchase a phone from the same firm. Market-dominant equilibria arise when default-switching costs are sufficiently small that the advantaged firm finds it more profitable to set the relatively low price required to attract all consumers than to set a higher price that would only attract close consumers. Proposition 2 characterizes outcomes in the market-dominant equilibrium in which all consumers purchase a phone from Firm 1. The proposition refers to the following conditions.

²⁷It can be shown that the first inequality in Condition 1B holds if and only if $[t - K_1] [2t + r_H - r_L] \leq [t + A]^2$. The second inequality in Condition 1B holds if and only if $[t - K_2] [2t - r_H + r_L] \leq [t - A]^2$.

²⁸Recall that a firm's "close" consumers are those with the strongest preferences for the firm's default PD setting, whereas a firm's "distant" consumers as those with the strongest preferences for the rival's default PD setting.

Condition 2A. $c_2 > r_H$. Condition 2B. $\Omega_1 \equiv \frac{1}{2} [r_H - r_L] - 3A - \frac{1}{8t} [t - A]^2 > 0$. Condition 2C. $K_1 \leq \frac{2t}{2t + r_H - r_L} \Omega_1$. Condition 2D. $K_1 < c_1 - r_L - 3A$.

Condition 2A ensures that $c_2 - r_H$, the minimum price that Firm 2 can profitably charge when it serves all consumers and no consumer changes the default PD setting, is positive. Conditions 2B and 2C ensure that Firm 1 secures positive profit when it attracts all consumers by setting p_1 marginally below $p_2 + G_1 - G_2 - K_1$ when Firm 2 sets $p_2 = c_2 - r_H$.²⁹ Condition 2D ensures that the highest value of p_1 that allows Firm 1 to attract all consumers when Firm 2 sets $p_2 = c_2 - r_H$ is positive.³⁰

Proposition 2. Suppose $K_1 \in (0, t)$, $K_2 \in (0, t)$, A < 0, and Conditions 2A - 2D hold. Then an equilibrium exists in which all consumers purchase a phone from Firm 1. At this equilibrium, all consumers located in $(\frac{1}{2} + \frac{K_1}{2t}, 1]$ (and only these consumers) change the default setting on the phone they purchase. Furthermore, p_1 is marginally below $c_1 - r_L - 3A - K_1 > 0$; $p_2 = c_2 - r_H > 0$; $\pi_1 \approx \frac{t - K_1}{2t} [r_H - r_L] - 3A - K_1 > 0$; and $\pi_2 = 0.^{31}$

To explain Proposition 2, it is helpful to identify two critical levels of advertising revenue:

$$r_{1} \equiv \frac{1}{2} [r_{L} + r_{H}] - \frac{K_{1}}{2t} [r_{H} - r_{L}]; \quad r_{2} \equiv \frac{1}{2} [r_{L} + r_{H}] + \frac{K_{2}}{2t} [r_{H} - r_{L}] > r_{1}.$$
(4)

 r_i is the revenue that Firm $i \in \{1, 2\}$ receives from advertisers when $K_i < t$ and all consumers purchase a phone from Firm *i*. Recall from (2) that a consumer who purchases a phone from Firm 1 will change the default setting on the phone if and only if she is located in $(\frac{1}{2} + \frac{K_1}{2t}, 1]$. Therefore, Firm 1's revenue from advertisers when $K_1 < t$ and all consumers buy a phone from Firm 1 is $r_L \left[\frac{1}{2} + \frac{K_1}{2t} \right] + r_H \left[1 - \left(\frac{1}{2} + \frac{K_1}{2t} \right) \right] = r_1$. Analogous considerations explain the expression for r_2 in (4).

Firm 2 must charge $p_2 \ge c_2 - r_2$ to profitably serve all consumers. Under the conditions specified in Proposition 2, Firm 1 can profitably attract all consumers by marginally undercutting any $p_2 \ge c_2 - r_2$. Firm 2 can profitably reduce p_2 below $c_2 - r_2$ if it does

 $^{^{29}}p_2 = c_2 - r_H$ is the lowest price that Firm 2 can profitably charge for its phone when every consumer that purchases a phone from Firm 2 retains the phone's default PD setting. (1) and (2) imply that when $K_1 < t$, the consumer located at 1 (and thus all consumers) will purchase a phone from Firm 1 rather than from Firm 2 if $G_1 - p_1 - K_1 > G_2 - p_2 \Leftrightarrow p_1 < p_2 + G_1 - G_2 - K_1$.

³⁰This value of p_1 is just below $c_2 - r_H + G_1 - G_2 - K_1 = c_1 - r_L + G_1 - c_1 + r_L - (G_2 - c_2 + r_H) - K_1 = c_1 - r_L - 3A - K_1$.

³¹Here and throughout the analysis, our reference to a price that is marginally below a specified level "X" should be interpreted as the highest feasible price that is strictly less than X. Technically, this price does not exist when feasible prices are only required to be positive. However, this price would exist if attention were restricted to a finite number of (positive) prices. This restriction is not imposed, for analytic and expositional ease.

not serve consumers located in $[0, \frac{1}{2} - \frac{K_2}{2t}]$. In particular, Firm 2 can profitably reduce p_2 as low as $c_2 - r_H$ if it only serves consumers in $[\frac{1}{2} - \frac{K_2}{2t}, 1]$. (Recall from (3) that these consumers do not change the default PD setting on a phone they purchase from Firm 2.) Therefore, Firm 2 will reduce p_2 to $c_2 - r_H$ to counteract an attempt by Firm 1 to attract all consumers by setting p_1 marginally below $p_2 + G_1 - G_2 - K_1$. In the unique equilibrium in which Firm 1 serves all consumers, Firm 2 sets $p_2 = c_2 - r_H$ and Firm 1 sets p_1 marginally below $p_2 + G_1 - G_2 - K_1$.

Proposition 3 characterizes the outcomes that arise in the market-dominant equilibria in which all consumers purchase a phone from Firm 2. The proposition refers to the following conditions.

Condition 3A. $c_1 > r_1$. Condition 3B. $K_2 < G_2 - G_1 + c_1 - r_1$. Condition 3C. $K_2 \left[\frac{2t - r_H + r_L}{2t} \right] < \Omega_2(p_1)$ for all $p_1 \in [c_1 - r_1, \min\{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$, where $\Omega_2(p_1) \equiv p_1 + G_2 - G_1 - c_2 + \frac{1}{2} [r_H + r_L] - x_2(p_1)$, and $x_2(p_1) \equiv \frac{1}{2t} [t + G_2 - G_1 - c_2 + r_H + p_1]^2 \ge 0$.

Condition 3A ensures that $c_1 - r_1$, the minimum price that Firm 1 can profitably charge when $K_1 < t$ and all consumers purchase a phone from Firm 1, is positive. Condition 3B ensures that the highest p_2 that induces all consumers to purchase a phone from Firm 2 when Firm 1 sets $p_1 = c_1 - r_1$ is positive.³³ Condition 3C ensures that Firm 2 secures positive profit when it successfully attracts all consumers by setting p_2 marginally below $p_1 + G_2 - G_1 - K_2$ when $p_1 = c_1 - r_L$.

Proposition 3. Suppose $K_1 \in (0, t)$, $K_2 \in (0, t)$, A > 0, and Conditions 3A - 3C hold. Then a family of equilibria exist in which all consumers purchase a phone from Firm 2. In each of these equilibria, all consumers located in $\left[0, \frac{1}{2} - \frac{K_2}{2t}\right)$ (and only these consumers) change the default setting on the phone they purchase. Furthermore, $p_1 \in [c_1 - r_1, \min\{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}] > 0$; p_2 is marginally below $p_1 + G_2 - G_1 - K_2 > 0$; $\pi_1 = 0$; and $\pi_2 \approx p_1 - c_2 + G_2 - G_1 - K_2 + \frac{1}{2}[r_H + r_L] + \frac{K_2}{2t}[r_H - r_L] > 0$.

³²Recall that we restrict attention to undominated strategies for Firms 1 and 2. When identifying equilibrium strategies, we also assume that a firm will not change its price if it has no strict incentive to do so.

³³This highest price is $c_1 - r_1 + G_2 - G_1 - K_2$. This is the case because (3) implies that when $K_2 < t$, the consumer located at 0 (and thus all consumers) will prefer to purchase a phone from Firm 2 rather than from Firm 1 if $G_2 - p_2 - K_2 > G_1 - p_1 \Leftrightarrow p_2 < p_1 + G_2 - G_1 - K_2$.

The multiplicity of equilibria identified in Proposition 3 reflect the following considerations. Suppose Firm 1 initially sets price $\hat{p}_1 \in [c_1 - r_1, c_1 - r_L]$. Then Firm 2 can set p_2 marginally below $\hat{p}_2 = \hat{p}_1 + G_2 - G_1 - K_2$, and thereby ensure that all consumers purchase a phone from Firm 2. In response, Firm 1 could conceivably reduce p_1 marginally in an attempt to attract consumers. However, doing so would only attract consumers in $[0, \frac{1}{2} - \frac{K_2}{2t}]$, who generate the lowest advertising payment, r_L . Firm 1 cannot profitably serve these consumers with a price below $c_1 - r_L$.

Conceivably, Firm 1 might consider a more pronounced reduction in p_1 in an attempt to attract all consumers, including those who generate the highest advertising revenue, r_H . Firm 1 would have to set $p_1 \leq p_2 + G_1 - G_2 - K_1$ to do so.³⁴ However, such low prices are not profitable for Firm 1 when $p_2 + G_1 - G_2 - K_1 < c_1 - r_1$.

In summary, when $p_1 = \hat{p}_1 \in (c_1 - r_1, c_1 - r_L]$ and Firm 2 sets p_2 marginally below $\hat{p}_2 = \hat{p}_1 + G_2 - G_1 - K_2$, Firm 2 will attract all consumers. Firm 1 cannot profitably reduce p_1 marginally because doing so would primarily attract consumers who only generate advertising revenue r_L . Furthermore, Firm 1 cannot profitably reduce p_1 by an amount sufficient to attract all consumers when $p_2 + G_1 - G_2 - K_1 < c_1 - r_1$. Therefore, Firm 1 has no strict incentive to change $\hat{p}_1 \in (c_1 - r_1, c_1 - r_L]$ under these conditions. Consequently, multiple equilibria arise in which all consumers purchase a phone from Firm 2.³⁵

5 The Effects of Default-Switching Costs on Profits

Having characterized the key features of market-sharing (MS) and market-dominant (MD) equilibria and the conditions under which they arise, we now examine how defaultswitching costs affect firms' profits in these equilibria. Proposition 4 reports that the impact of default-switching costs on profits can vary across firms and across equilibria. The proposition refers to: (i) $\pi_j^{Di}(K_1, K_2)$, which is Firm j's profit, given K_1 and K_2 , in a MD*i* equilibrium, which is a MD equilibrium in which all consumers purchase a phone from Firm $i \in \{1, 2\}$; (ii) $\underline{\pi}_j^{Di}(K_1, K_2)$, which is the minimum profit that Firm j secures in MD*i* equilibria; and (iii) π_i^S , which is Firm *i*'s profit in the MS equilibrium characterized in Proposition 1.

³⁴Recall that (1) and (2) imply that a consumer located at 1 (and thus all consumers) will purchase a phone from Firm 1 when $K_1 < t$ if $G_1 - p_1 - K_1 > G_2 - p_2 \Leftrightarrow p_1 < p_2 + G_1 - G_2 - K_1$.

³⁵In contrast, a unique equilibrium arises in the setting of Proposition 2, where Firm 1 is the advantaged firm. In this case, Firm 2 find it profitable to marginally undercut any p_1 above $c_2 - r_H$ at which Firm 1 is initially serving all cosnumers. This is the case because a marginal reduction in p_2 below p_1 enables Firm 2 to profitably attract close consumers who generate the high level of advertising revenue, r_H .

Proposition 4. (i) $\frac{\partial \pi_1^{D_1}(\cdot)}{\partial K_1} < 0$ and $\frac{\partial \pi_1^{D_1}(\cdot)}{\partial K_2} = 0$ in the MD1 equilibrium characterized in Proposition 2; (ii) $\frac{\partial \pi_2^{D_2}(\cdot)}{\partial K_2} \gtrless 0 \Leftrightarrow r_H - r_L \gtrless 2t$ and $\frac{\partial \pi_2^{D_2}(\cdot)}{\partial K_1} > 0$ in the MD2 equilibria characterized in Proposition 3; and (iii) $\frac{\partial \pi_1^{S_1}(\cdot)}{\partial K_1} = \frac{\partial \pi_1^{S_1}(\cdot)}{\partial K_2} = \frac{\partial \pi_2^{S_2}(\cdot)}{\partial K_2} = \frac{\partial \pi_2^{S_2}(\cdot)}{\partial K_1} = 0$ in the MS equilibrium characterized in Proposition 1.

Conclusion (i) in Proposition 4 indicates that when Firm 1 is the advantaged firm, its profit declines as its default-switching cost (K_1) increases in a MD1 equilibrium for two reasons. First, to continue to attract all relatively distant consumers (i.e., those located in $(\frac{1}{2} + \frac{K_1}{2t}, 1])$ who change the default PD setting on the phone they purchase from Firm 1, Firm 1 must reduce p_1 sufficiently to offset the higher default-switching cost that these customers experience. Second, the increase in K_1 induces more customers to retain Firm 1's default PD setting rather than switch to their preferred PD setting. Firm 1 receives a lower payment from advertisers (r_L rather than r_H) for every customer that retains Firm 1's default "privacy" setting rather than switching to the "disclosure" setting.

Conclusion (i) in Proposition 4 also indicates that Firm 1's profit is unaffected by changes in Firm 2's default-switching cost in the MD1 equilibrium identified in Proposition 2. This is the case because Firm 2 sets $p_2 = c_2 - r_H$ in this equilibrium, which is the lowest price that allows Firm 2 to secure nonnegative profit when it serves close customers who do not change the default PD setting on Firm 2's phone. Because this price does not vary with K_2 , the price that Firm 1 must set to attract all consumers does not vary with K_2 .³⁶

In contrast, conclusion (ii) in Proposition 4 reports that Firm 2's profit can increase as the rival's default-switching cost (K_1) increases in the MD2 equilibria characterized in Proposition 3. This is the case because the lowest price that allows Firm 1 to secure nonnegative profit when it serves all consumers is $c_1 - r_1$. Recall that r_1 is the average per-customer advertising revenue that Firm 1 secures when it serves all consumers. This average advertising revenue declines as K_1 increases because more consumers retain Firm 1's default "privacy" setting (which generates the smaller advertising revenue, r_L , for Firm 1) as K_1 increases.³⁷ The reduction in its advertising revenue compels Firm 1 to increase p_1 to avoid negative profit. The increase in the lowest price that Firm 1 will charge enables Firm 2 to continue to attract all consumers with a higher price, which increases Firm 2's equilibrium profit.³⁸

³⁶Recall from Proposition 2 that this price is approximately $p_1 = p_2 + G_1 - G_2 - K_1 = c_2 - r_H + G_1 - G_2 - K_1$ = $c_1 - r_L - 3A - K_1$.

³⁷Formally, (4) implies that $\frac{\partial r_1}{\partial K_1} = \frac{\partial}{\partial K_1} \left(\frac{1}{2} \left[r_L + r_H \right] - \frac{K_1}{2t} \left[r_H - r_L \right] \right) = -\frac{r_H - r_L}{2t} < 0.$

³⁸Firm 2's maximum profit in the MD equilibria characterized in Proposition 3 also increases as K_1 increases if $c_1 - r_1 + K_1 + K_2 < c_1 - r_L$. Firm 2's profit increases with K_1 in this case in part because the higher default-switching costs that distant purchasers of Firm 1's phone would incur enable Firm 2 to attract close consumers with a higher price.

Conclusion (ii) in Proposition 4 also reports that an increase in K_2 can either increase or reduce Firm 2's minimum profit in the MD2 equilibria characterized in Proposition 3. This is the case because an increase in K_2 has two countervailing effects on $\underline{\pi}_2^{D2}(\cdot)$. First, an increase in K_2 reduces the utility that consumers located in $[0, \frac{1}{2} - \frac{K_2}{2t})$ derive from Firm 2's phone because these consumers now incur a higher cost when they change the default PD setting on the phone they purchase from Firm 2 to their preferred "privacy" setting. This reduced utility requires a reduction in p_2 to ensure the continued patronage of these consumers, which reduces Firm 2's profit, *ceteris paribus*.

Second, the reduced switching away from Firm 2's default "disclosure" setting increases the payment that Firm 2 receives from advertisers. As conclusion (ii) in Proposition 4 reports, if the increase in per-customer advertising payment is sufficiently large (i.e., if $r_H - r_L > 2t$), then the reduced default-switching induced by an increase in K_2 increases Firm 2's minimum profit in the MD2 equilibria characterized in Proposition 3.

Conclusion (iii) in Proposition 4 indicates that changes in default-switching costs do not affect profits in the MS equilibrium characterized in Proposition 1. This conclusion reflects the fact that in this equilibrium, each firm sets a relatively high price and thereby secures a relatively high profit margin on close customers, while effectively ceding distant customers to the rival supplier. In particular, the consumer who is indifferent between purchasing a phone from Firm 1 and Firm 2 strictly prefers to retain the default PD setting on the phone she purchases.³⁹ Consequently, a marginal reduction in default-switching costs would not induce any consumer to change the default PD setting on the phone she purchases in this equilibrium. Therefore, the firms' profits do not change as default-switching costs change in this equilibrium.

6 Comparing Profits Across Equilibria

Having examined how profits vary with switching costs in MD and MS equilibria, we now compare the firms' profits in MD and MS equilibria. Doing so will allow us to identify conditions under which the advantaged firm would benefit from a unilateral increase in its default-switching cost. The ensuing analysis refers to the following specific magnitudes of firms' advantages and transportation costs.

$$|A|_{1L} \ \equiv \ 2\,t - \sqrt{3\,t^2 + 2\,t\,\Delta} \ . \qquad |A|_{1H} \ \equiv \ 2\,t + \sqrt{3\,t^2 + 2\,t\,\Delta} \ .$$

³⁹Recall that $x_0 = \frac{1}{2} - \frac{A}{2t}$, $K_2 > A$, and $K_1 > |A|$ in the MS equilibrium characterized in Proposition 1. Consequently, $x_0 \in (\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2} + \frac{K_1}{2t})$, so the consumer who is indifferent between purchasing a phone from Firm 1 and Firm 2 is located in the region where no consumer will change the default PD setting on the phone she purchases.

$$A_{2L} \equiv 2t - \sqrt{3t^2 - 2t\Delta} \,. \qquad A_{2H} \equiv 2t + \sqrt{3t^2 - 2t\Delta} \,.$$
$$t_{1L} \equiv \Delta + 2|A| - \sqrt{3|A|^2 + 4|A|\Delta + \Delta^2} \,. t_{1H} \equiv \Delta + 2|A| + \sqrt{3|A|^2 + 4|A|\Delta + \Delta^2} \,.$$
$$t_{2L} \equiv 2A - \Delta - \sqrt{3A^2 - 4A\Delta + \Delta^2} \,. t_{2H} \equiv 2A - \Delta + \sqrt{3A^2 - 4A\Delta + \Delta^2} \,.$$
(5)

Lemma 3. Suppose the conditions in Proposition 1 hold and A < 0. Then $\pi_1^S > \pi_1^{D1}(0,0)$ if $t > t_{1H}$, whereas $\pi_1^S < \pi_1^{D1}(0,0)$ if $t \in (K_1, t_{1H})$. Furthermore, if $t > 2\Delta$, then $\pi_1^S > \pi_1^{D1}(0,0)$ if $|A| < |A|_{1L}$, whereas $\pi_1^S < \pi_1^{D1}(0,0)$ if $|A| \in (|A|_{1L}, t)$. If $t \le 2\Delta$, then $\pi_1^S < \pi_1^{D1}(0,0)$ if $|A| \in (|A|_{1L}, t)$. If $t \le 2\Delta$, then $\pi_1^S < \pi_1^{D1}(0,0)$ for all $|A| \in (0,t)$.

Lemma 4. Suppose the conditions in Proposition 1 hold and $A > \Delta$. Then: (i) $\pi_2^S > \pi_2^{D2}(0,0)$ if $t > t_{2H}$ or $A < A_{2L}$; and (ii) $\pi_2^S < \pi_2^{D2}(0,0)$ if $t \in (K_2, t_{2H})$ or $A \in (A_{2L}, t)$.

Lemmas 3 and 4 identify conditions under which the advantaged firm secures more profit in the MS equilibrium that entails positive default-switching costs for the advantaged firm than in the MD equilibrium with no default-switching costs. The lemmas report that the advantaged firm benefits from higher switching costs when its advantage (|A|) is sufficiently limited or the extent of horizontal product differentiation (t) is sufficiently pronounced.

These conclusions reflect the fact that when a firm enjoys only a limited advantage over its rival, the firm must set a relatively low price to attract all consumers, which limits the profit the firm secures in the MD equilibrium. The firm with a limited advantage can earn more profit in the MS equilibrium when a relatively high default-switching cost makes it relatively unprofitable for the firm to attract distant consumers, so the firm sets a relatively high price and serves only close consumers.⁴⁰

When the extent of horizontal product differentiation (t) is relatively pronounced and default-switching costs are relatively large (i.e., $K_1 > |A|$ and $K_2 > A$, as in Proposition 1), consumers are willing to pay a relatively high price for a phone with the consumer's preferred default PD setting. Consequently, when t is relatively large, the advantaged firm can secure more profit by setting a relatively high price and attracting only close customers than by

⁴⁰When a relatively high default-switching cost prevails, a distant consumer will only purchase a phone with her less-preferred default PD setting if the price of the phone is relatively low. This is the case because the consumer recognizes before she purchases the phone that she will ultimately incur the relatively high default-switching cost to secure her preferred PD setting.

setting the lower price required to attract all consumers.⁴¹

Lemmas 3 and 4 help to identify conditions under which, if default-switching costs were endogenous, a firm could enhance its profit by unilaterally increasing its default-switching cost, even in the absence of any cost savings from doing so. In this sense, the firm could benefit from engaging in "self-sabotage" that renders its product less attractive to some consumers without increasing the product's appeal to any consumer. Proposition 5 specifies conditions under which Firm 2 can benefit from such self-sabotage when it is the advantaged firm and default-switching costs are initially 0.

Proposition 5. Suppose: (i) $G_2 - c_2 > G_1 - c_1$; (ii) max $\{\Delta, r_H - c_2 - t\} < A < \min\{t, t + c_1 - r_L\}$;⁴² (iii) $t > t_{2H}$; and (iv) $K_1 = 0$. Then Firm 1 and Firm 2 both secure strictly greater profit in equilibrium when $K_2 \in (A, t)$ than when $K_2 = 0$.

To explain in more detail why an increase in the default-switching cost of the advantaged firm (Firm 2) can increase the equilibrium profit of both industry suppliers, recall that when a consumer can costlessly change the default PD setting on the phone she purchases, the consumer perceives the phones to exhibit no horizontal product differentiation. In such a setting, the firms compete relatively aggressively to serve all consumers. This intense price competition can keep Firm 2's equilibrium profit relatively low, even though $G_2 - c_2 >$ $G_1 - c_1$.⁴³

By increasing K_2 , Firm 2 ensures that consumers located in $(\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2})$ consider the products of Firms 1 and 2 to be horizontally differentiated. This is the case because the switching cost, $K_2 > 0$, deters these consumers from switching to their preferred PD setting ("privacy") if they buy a phone from Firm 2. Consequently, by increasing K_2 , Firm 2 expands the region in which its competition with Firm 1 is effectively "softened" by perpetuating the default horizontal product differentiation.

The softened competition reflects in part the following consideration. By increasing K_2 , Firm 2 reduces the utility that consumers in $(\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2})$ derive from purchasing Firm 2's phone (because these consumers will no longer switch to their preferred privacy setting if they purchase a phone from Firm 2). The associated reduced willingness to pay for Firm 2's phone implies that Firm 2 must reduce its price to attract these consumers. Firm 2 may

⁴¹The Appendix presents four corollaries to Lemmas 3 and 4. The corollaries explain how the range of settings in which default-switching costs increase the profit of the advantaged firm by inducing a MS equilibrium varies with model parameters.

⁴²Recall that $\Delta \equiv r_H - r_L > 0$.

⁴³Recall from Lemma 2 that when $K_1 = K_2 = 0$ and Firm 2 is the advantaged firm, Firm 2's equilibrium profit is $G_2 - c_2 - (G_1 - c_1)$. This profit approaches 0 as $G_2 - c_2$ approaches $G_1 - c_1$.

find it unprofitable to do so because the price reduction applies to all consumers that Firm 2 attracts, not only those in $(\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2})$. Consequently, the increase in K_2 can effectively endow Firm 2 with a commitment to set a relatively high price that only attracts close customers, but generates a relatively high profit margin for Firm 2 on each phone that it sells. Because prices are strategic complements, Firm 2's relatively high price induces Firm 1 to set a relatively high price. This reduced pricing aggression can increase the equilibrium profit of both firms.

An increase in K_2 can deliver an additional benefit to Firm 2. As K_2 increases, fewer consumers who purchase Firm 2's phone change the default PD setting on their phone. Consequently, Firm 2 secures more advertising revenue (r_H rather than r_L) for each phone that it sells to a consumer who no longer changes Firm 2's default PD setting.

Conditions (ii) and (iii) in Proposition 5 require t to be sufficiently large. When t is large, the softening of competition in $(\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2})$ is relatively pronounced because each firm now has substantial market power over close consumers. Consequently, it becomes relatively unprofitable for a firm to attract distant consumers. The higher equilibrium prices that arise under softened competition increase Firm 2's profit.⁴⁴

Proposition 6 provides a corresponding conclusion in the setting where Firm 1 is the advantaged firm.

Proposition 6. Suppose: (i) A < 0; (ii) $r_H - c_2 - t < A < t + c_1 - r_L$; (iii) t > |A|; (iv) $t > t_{1H}$; and (v) $K_2 = 0$. Then Firm 1 and Firm 2 both secure strictly greater profit in equilibrium when $K_1 \in (|A|, t)$ than when $K_1 = 0$.

The explanation for Proposition 6 parallels the explanation for Proposition 5 with one exception. In addition to reducing the attraction of its phone to relatively distant customers who switch the default PD setting, an increase in K_1 that expands the set of customers who do not switch the default PD setting reduces Firm 1's advertising revenue (because $r_L < r_H$). This consideration implies that Firm 1 is less likely to benefit from a unilateral increase in its default-switching cost than Firm 2 in the sense that condition (iv) in Proposition 6 is more restrictive than condition (iii) in Proposition 5.⁴⁵

In summary, an increase in K_i has three key effects. First, it reduces the number of the most distant potential customers who will switch the default PD setting on a phone purchased from Firm *i*. Second, it increases the number of the less distant customers who will retain the

⁴⁴Firm 1's profit also increases because the disadvantaged firm secures no profit in a MD equilibrium.

⁴⁵It is readily verified that $t_{1H} > t_{2H}$ when the |A| in t_{1H} and the A in t_{2H} have the same magnitude.

default PD setting on a phone purchased from Firm i.⁴⁶ Third, it reduces the amount that the most distant consumers are willing to pay for Firm *i*'s phone (because they anticipate incurring the (increased) default-switching cost). These three effects imply that an increase in K_i reduces the attraction to Firm *i* of competing for the most distant potential customers (given their reduced number and their reduced willingness to pay for Firm *i*'s phone).⁴⁷ This reduced attraction can serve as a credible commitment not to compete aggressively for the entire market (and for the most distant customers, in particular), which can induce the rival supplier to compete less aggressively in return.

7 The Setting with Endogenous K

We now extend the foregoing analysis by allowing default-switching costs to be endogenous. Formally, we examine equilibrium outcomes in the *setting with endogenous* K, where the interaction between Firm 1 and Firm 2 proceeds in two stages. In the first stage, Firm 1 chooses K_1 and Firm 2 chooses K_2 , simultaneously and noncooperatively. In the second stage, Firm 1 chooses p_1 and Firm 2 chooses p_2 , simultaneously and noncooperatively. For simplicity, we abstract from any costs that a firm might incur to change the default-switching cost on its phone.

Proposition 7 identifies conditions under which the market-sharing (MS) equilibrium characterized in Proposition 1 persists in the setting with endogenous K. The conditions include t being sufficiently large relative to |A|. The relatively large transportation cost implies that a firm must set a relatively low price to secure the patronage of all consumers, even after reducing its default-switching cost to 0. Consequently, rather than eliminating its default-switching cost and competing to attract all consumers, each firm prefers to implement a relatively high default-switching cost and charge a relatively high price that serves to attract (only) relatively close consumers in a MS equilibrium.

Proposition 7. Suppose: (i) (K_1^*, K_2^*) are such that conditions (ii) and (iii) in Proposition 1 hold; (ii) t > |A|; (iii) Condition 1A holds; (iv) Condition 1B holds when $K_1 = K_2 = 0$; and (v) $\Delta < 2t$. Then (K_1^*, K_2^*) , along with $p_1 = c_1 - r_L + t - A$ and $p_2 = c_2 - r_H + t + A$, constitute a MS equilibrium in the setting with endogenous K.

Propositions 8 and 9 identify conditions under which market-dominant equilibria with 0 default-switching costs arise in the setting with endogenous K. When a firm's advantage

⁴⁶Distant customers for Firm 1 here are those located in $(\frac{1}{2}, 1]$. The most distant of these customers are those located in $(\frac{1}{2}, \frac{1}{2t}, \frac{K_1}{2t}, 1]$. The less distant of these customers are those located in $(\frac{1}{2}, \frac{1}{2}, \frac{K_1}{2t}, 1]$.

⁴⁷The reduced attraction is particularly pronounced for Firm 2 because its most distant potential customers switch away from the relatively lucrative default PD setting if they purchase a phone from Firm 2.

is sufficiently pronounced relative to t, the firm can secure greater profit by eliminating its default-switching cost and competing to attract all consumers than by adopting a defaultswitching cost that gives rise to a MS equilibrium. When $\Delta < 2t$, the advertising revenue that Firm 2 foregoes by allowing customers to costlessly switch the default PD setting on its phone is less than the extra profit the firm secures from the higher price that distant consumers will pay for Firm 2's phone when they can costlessly implement their preferred PD setting on the phone.⁴⁸

Proposition 8. Suppose A < 0, $G_1 - c_1 - (G_2 - c_2) > \max\{\frac{\Delta}{2}, \frac{1}{2t}[t + |A|]^2\}$, $c_2 > \frac{1}{2}[r_L + r_H]$, and $c_1 > r_L$. Then in the setting with endogenous K, there exists a MD1 equilibrium in which: (i) $K_1 = K_2 = 0$; (ii) $p_2 = c_2 - \frac{1}{2}[r_L + r_H]$; and (iii) p_1 is marginally below $c_2 - r_H + G_1 - G_2$.

Proposition 9. Suppose A > 0, $G_2 - c_2 - (G_1 - c_1) > \frac{1}{2t} [t + A]^2$, $c_2 > r_H$, $c_1 > \frac{1}{2} [r_L + r_H]$, and $\Delta < 2t$. Then in the setting with endogenous K, there exists a MD2 equilibrium in which: (i) $K_1 = K_2 = 0$; (ii) $p_1 = c_1 - \frac{1}{2} [r_L + r_H]$; and (iii) p_2 is marginally below $c_1 - \frac{1}{2} [r_L + r_H] + G_2 - G_1$.

Proposition 10 reports that when $\Delta > 2t$, Firm 2 can prefer to compete to attract all consumers by raising K_2 sufficiently to ensure that no customer switches the default PD setting on Firm 2's phone. By doing so, Firm 2 secures a larger increase in advertising revenue than the reduction in sales revenue it incurs when it must reduce p_2 to compensate its distant customers for retaining their less-preferred PD setting on the phone they purchase from Firm 2.

Proposition 10. Suppose: (i) $K_1 = 0$ and $K_2 = K_2^* \ge t$; (ii) Condition 3A holds; (iii) $t < G_2 - G_1 + c_1 - r_1$; (iv) $\frac{2t - \Delta}{2} < \Omega_2(p_1)$ for all $p_1 \in [c_1 - r_1, \min\{c_1 - r_L, c_1 - r_1 + K_1 + t\}]$,⁴⁹ (v) $G_2 - c_2 > G_1 - c_1$; and (vi) $\Delta > 2t$. Then in the setting with endogenous K, $(0, K_2^*)$ and (p_1, p_2) prices such that $p_1 \in [c_1 - r_1, \min\{c_1 - r_L, c_1 - r_1 + t\}]$ and p_2 is marginally below $p_1 + G_2 - G_1 - t$ constitute a MD2 equilibrium.

Before concluding, we demonstrate that some equilibria that can arise when K_1 and K_2 are exogenous cannot exist in the setting with endogenous K. Proposition 11 identifies

⁴⁸Recall from Proposition 4 that Firm 1's profit when it serves all consumers declines with K_1 . An increase in K_1 both reduces the attraction of Firm 1's phone to distant consumers and reduces Firm 1's advertising revenue (because $r_L < r_H$).

⁴⁹Recall that Ω_2 is defined by Condition 3C.

conditions under which a MD1 equilibrium in which $K_1 > 0$ does not exist. In any such putative equilibrium, Firm 1 could increase its profit by reducing K_1 . The reduction in K_1 would allow Firm 1 to: (i) increase its revenue by attracting distant customers with a higher p_1 ; and (ii) increase the payments that Firm 1 secures from advertisers by inducing more distant customers to change the default "privacy" setting on the phone they purchase from Firm 1.

Proposition 11. Suppose (i) A < 0; (ii) t < |A|; (iii) $G_1 - c_1 - (G_2 - c_2) > \frac{\Delta}{2}$; (iv) Conditions 2A and 2B hold; (v) $t < c_1 - r_L - 3A$; (vi) and (vii) $t \le \frac{2t}{2t + r_H - r_L} \Omega_1$.⁵⁰ Then a MD1 equilibrium in which $K_1 > 0$ and $K_2 \ge 0$ does not exist in the setting with endogenous K.

Proposition 12 identifies conditions under which a MD2 equilibrium in which $K_2 > 0$ does not exist when $\Delta < 2t$. In any such putative equilibrium, Firm 2 could increase its profit by reducing K_2 . The smaller default-switching cost would allow Firm 2 to attract distant consumers with a higher p_2 . When $\Delta < 2t$, the associated increase in revenue exceeds the reduction in payments that advertisers deliver to Firm 2 due to the increased switching of the default "disclosure" setting on Firm 2's phone.

Proposition 12. Suppose (i) $G_2 - c_2 - (G_1 - c_1) > \frac{\Delta}{2}$; (ii) $2t > \Delta$; (iii) $c_1 > \frac{1}{2} [r_H + r_L]$; (iv) $t < G_2 - G_1 + c_1 - \frac{1}{2} [r_H + r_L]$; (v) Condition 3C holds if $t < K_2$; and (vi) $\frac{1}{2} [2t - \Delta] < \Omega_2(p_1)$ for all $p_1 \in [c_1 - r_1, \min \{c_1 - r_L, c_1 - r_1 + K_1 + t\}]$ if $t \geq K_2$.⁵¹ Then a MD2 equilibrium in which $K_1 \in [0, t), K_2 > 0$, and $p_1 = c_1 - r_1$ does not exist in the setting with endogenous K.

Finally, Proposition 13 identifies conditions under which a MD2 equilibrium in which $K_2 < t$ does not exist when $\Delta > 2t$. In any such putative equilibrium, Firm 2 could increase its profit by increasing K_2 . The higher default-switching cost would require Firm 2 to reduce p_2 to continue to attract distant consumers. However, when $\Delta > 2t$, the associated reduction in revenue is outweighed by the increase in payments that advertisers deliver to Firm 2 due to the reduced switching of the default "disclosure" setting on Firm 2's phone.

Proposition 13. Suppose (i) $G_2 - c_2 > G_1 - c_1$; (ii) $\Delta > 2t$; and (iii) Conditions 3A - 3C hold. Then a MD2 equilibrium in which $K_1 \in [0, t)$, $K_2 \in [0, t)$, and $p_1 = c_1 - r_1$ does not exist in the setting with endogenous K.

⁵⁰Recall that Ω_1 is defined in Condition 2B.

⁵¹Recall that $\Omega_2(p_1)$ is defined in Condition 3C.

8 Conclusions

We have analyzed a streamlined model of Hotelling competition between two suppliers of smartphones. Our model differs from the standard model of Hotelling competition primarily by allowing customers to change the default privacy/disclosure (PD) setting on the phone they purchase. When consumers can costlessly change this setting, horizontal product differentiation is effectively eliminated. Relatively intense competition can ensue, giving rise to market dominant equilibria is which all consumers purchase a phone from the same supplier.

Non-trivial default-switching costs restore meaningful horizontal product differentiation, which can enhance supplier profit under certain conditions and can induce market sharing equilibria in which both suppliers serve consumers. The supplier with the default "disclosure" setting can be particularly likely to benefit from higher default-switching costs because the reduced default switching that occurs in equilibrium ensures that the supplier secures higher payments from advertisers. Our findings suggest one reason why suppliers might decline to implement the levels of default-switching costs that best serve consumers, and why suppliers might sometimes be inclined to frustrate regulatory efforts to reduce industry switching costs.

Our finding that the supplier with the default "disclosure" setting can be relatively likely to benefit from an increase in its default-switching costs is consistent with observed industry practice. Changing the default PD setting is generally thought to be more onerous on Samsung Galaxy phones, for example, than on iPhones. An iPhone user need only respond to a prompt that appears automatically when the user launches an app. In contrast, a Samsung Galaxy user must locate and choose the relevant option in the Settings menu on the phone. The user is not automatically prompted to do so.⁵²

Our model was intentionally streamlined to facilitate both a tractable analysis and a focus on the effects of default switching. Future research might consider several extensions of our model. Specifically, additional dimensions of consumer heterogeneity might be admitted. When consumers have different incomes or different innate valuations of smartphone services, for example, market sharing equilibria may arise even in the absence of default-switching costs. Market sharing equilibria may also be relatively likely to arise in the presence of additional dimensions of horizontal product differentiation (e.g., phone size, shape, and color).⁵³

Future research might also consider endogenous default PD settings, default-switching

⁵²See Grant and *Bloomberg* (2021) and O'Flaherty (2021), for example. Athey et al. (2017) observe that small navigation frictions can substantially reduce the likelihood that users change default settings on smartphones.

⁵³Market sharing equilibria may also be relatively likely to arise if consumers are concentrated near the center of the [0, 1] interval, rather than uniformly distributed in this interval.

costs that vary across customers, supplier costs of reducing customer default-switching costs, and repeat purchases of ever-evolving smartphones.⁵⁴ These model extensions can help to assess the robustness of our findings and perhaps provide new insights about equilibrium outcomes when firms compete in settings where consumers can alter default levels of horizontal product differentiation.

⁵⁴Future research might also consider settings where, in contrast to the present smartphone application with a binary "privacy" or "disclosure" PD setting, it is more natural to view consumers as having a choice among several levels of a horizontal product characteristic.

Appendix

Part A of this Appendix presents four corollaries to Lemmas 3 and 4. Part B of this Appendix outlines the proofs of the formal conclusions in the text. Chakravorty and Sappington (2024) provides the proofs of the corollaries to Lemmas 3 and 4, along with detailed proofs of the formal conclusions in the text

A. Corollaries to Lemmas 3 and 4.

The following Corollaries to Lemmas 3 and 4 explain how the range of settings in which default-switching costs increase the profit of an advantaged firm by inducing a MS equilibrium varies with model parameters.

Corollary 3.1. $\frac{dt_{1H}}{d|A|} > 0$ and $\frac{dt_{1H}}{d\Delta} > 0$, so the range of t realizations for which defaultswitching costs increase Firm 1's profit in the setting of Lemma 3 (i.e., $t > t_{1H}$) contracts as |A| increases or as Δ increases.

Corollary 3.2. $\frac{d|A|_{1L}}{d\Delta} < 0$. Furthermore, $\frac{d|A|_{1L}}{dt} > 0$ if $\Delta < \frac{t}{2}$. Therefore, the range of |A| realizations for which default-switching costs increase Firm 1's profit in the setting of Lemma 3 (i.e., $|A| < |A|_{1L}$): (i) contracts as Δ increases; and (ii) expands as t increases if $\Delta < \frac{t}{2}$.

Corollary 4.1. $\frac{dt_{2H}}{d\Delta} < 0$ and $\frac{dt_{2H}}{dA} > 0$, so the range of t realizations for which defaultswitching costs increase Firm 2's profit in the setting of Lemma 4 (i.e., $t > t_{2H}$) expands as Δ increases or as A declines.

Corollary 4.2. $\frac{dA_{2L}}{dt} > 0$ and $\frac{dA_{2L}}{d\Delta} > 0$, so the range of A realizations for which defaultswitching costs increase Firm 2's profit in the setting of Lemma 4 (i.e., $A < A_{2L}$) expands as t increases or as Δ increases.

Corollaries 3.1 and 4.1 report that as a firm's advantage (|A|) increases, the range of t realizations $(t > t_{1H} \text{ or } t > t_{2H})$ in which default-switching costs increase the advantaged firm's profit contracts. This is the case because the increased advantage increases the firm's equilibrium profit more rapidly in the MD equilibrium where it serves all consumers than in the MS equilibrium where it serves fewer consumers.

Corollaries 3.2 and 4.2 report that the range of advantage levels for which defaultswitching costs increase the advantaged firm's profit expands as t increases if: (i) Firm 2 is the advantaged firm; or (ii) Firm 1 is the advantaged firm and $\Delta < \frac{t}{2}$. The first conclusion arises because an increase in horizontal product differentiation reduces the intensity of price competition in the MS equilibrium, where the firms effectively focus on attracting close consumers. The reduced competitive intensity serves to increase the profit of Firm 2, which receives the large payment from advertisers (r_H) for every phone it sells in the MS equilibrium. The positive impact of reduced competitive intensity outweighs the countervailing effect that Firm 2 must reduce its price to attract customers as t increases, ceteris paribus. The same net effect for Firm 1 prevails when Δ is sufficiently small, so the advertising revenue that Firm 1 receives for each phone that it sells in the MS equilibrium is not too much smaller than the corresponding revenue that Firm 2 secures.

Corollaries 4.1 and 4.2 report that as the difference in advertising payments ($\Delta \equiv r_H - r_L$) increases, the range of t realizations ($t > t_{2H}$) and the range of A realizations ($A < A_{2L}$) in which default-switching costs increase Firm 2's profit both expand. This is the case because Firm 2's profit increases relatively rapidly in the MS equilibrium as Δ increases due to an increase in r_H because every consumer that Firm 2 serves in the MS equilibrium generates advertising revenue r_H .

Corollaries 3.1 and 3.2 report that as Δ increases, the range of t realizations $(t > t_{1H})$ and the range of |A| realizations $(|A| < |A|_{1L})$ in which default-switching costs increase Firm 1's profit both contract. This is the case because Firm 1 secures advertising revenue r_L for every phone it sells in the MS equilibrium, and this revenue declines as Δ increases, holding r_H constant.

B. Proofs of Formal Conclusions in the Text.

The following lemmas (Lemmas A1 – A18) are employed to prove the formal conclusions in the text.⁵⁵

Lemma A1. A user who buys a phone from Firm 2 will change the default setting on the phone if and only if the user is located in $[0, \frac{1}{2} - \frac{K_2}{2t})$.

Lemma A2. A user who buys a phone from Firm 1 will change the default setting on the phone if and only if the user is located in $(\frac{1}{2} + \frac{K_1}{2t}, 1]$.

Lemma A3. A user located in $\left[\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2} + \frac{K_1}{2t}\right]$ will not change the default setting on the phone she purchases.

Lemma A4. Suppose a user located at $x_0 \in \left[\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2} + \frac{K_1}{2t}\right]$ is indifferent between buying a phone from Firm 1 and from Firm 2. Then: (i) all users located in $[0, x_0)$ will buy a phone from Firm 1; and (ii) all users located in $[x_0, 1]$ will buy a phone from Firm 2.

Lemma A5. Suppose a user located at $x_1 \in [0, \frac{1}{2} - \frac{K_2}{2t}]$ is indifferent between buying a phone from Firm 1 and from Firm 2. Then: (i) all users located in $[0, \frac{1}{2} - \frac{K_2}{2t}]$ are similarly indifferent; and (ii) all users located in $(\frac{1}{2} - \frac{K_2}{2t}, 1]$ will buy a phone from Firm 2.

⁵⁵Chakravorty and Sappington (2024) provides the proofs of Lemmas A1 - A18.

Lemma A6. Suppose a user located at $x_2 \in (\frac{1}{2} + \frac{K_1}{2t}, 1]$ is indifferent between buying a phone from Firm 1 and from Firm 2. Then: (i) all users located in $[\frac{1}{2} + \frac{K_1}{2t}, 1]$ are similarly indifferent; and (ii) all users located in $[0, \frac{1}{2} + \frac{K_1}{2t})$ will buy a phone from Firm 1.

Lemma A7. If $p_1 \ge p_2 + G_1 - G_2 + K_2$, then all users located in $[0, \frac{1}{2} - \frac{K_2}{2t})$ (weakly) prefer to buy a phone from Firm 2 than from Firm 1. The preference is strict if the inequality holds strictly.

Lemma A8. If $p_1 \ge p_2 + G_1 - G_2 + K_2$, then all users located in $\left[\frac{1}{2} - \frac{K_2}{2t}, 1\right]$ (weakly) prefer to buy a phone from Firm 2 than from Firm 1. The preference is strict if the inequality holds strictly.

Lemma A9. If $p_2 \ge p_1 + G_2 - G_1 + K_1$, then all users located in $(\frac{1}{2} + \frac{K_1}{2t}, 1]$ (weakly) prefer to buy a phone from Firm 1 than from Firm 2. The preference is strict if the inequality holds strictly.

Lemma A10. If $p_2 \ge p_1 + G_2 - G_1 + K_1$, then all users located in $[0, \frac{1}{2} + \frac{K_1}{2t}]$ (weakly) prefer to buy a phone from Firm 1 than from Firm 2. The preference is strict if the inequality holds strictly.

Assumption 1. $K_1 \in [0, t), K_2 \in [0, t), \text{ and } (K_1, K_2) \neq (0, 0).$

Lemma A11. Suppose Assumption 1 holds. Then in any equilibrium in which $p_1 - p_2 \in (G_1 - G_2 - K_1, G_1 - G_2 + K_2)$: (i) all users located in $[0, \frac{1}{2} - \frac{K_2}{2t})$ strictly prefer to buy a phone from Firm 1 than from Firm 2; (ii) all users located in $(\frac{1}{2} + \frac{K_1}{2t}, 1]$ strictly prefer to buy a phone from Firm 2 than from Firm 1; and (iii) some user located in $[\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2} + \frac{K_1}{2t}]$ is indifferent between buying a phone from Firm 1 and from Firm 2.

Lemma A12. When Assumption 1 holds: (i) $c_1 - r_1$ is the lowest price that Firm 1 can profitably charge when all users buy a phone from Firm 1; and (ii) $c_2 - r_2$ is the lowest price that Firm 2 can profitably set when all users buy a phone from Firm 2.

Lemma A13. Suppose Assumption 1 holds. Then $c_1 - r_L > c_1 - r_1$ and $c_2 - r_H < c_2 - r_2$.

Lemma A14. Suppose Assumption 1 holds. Then: (i) setting $p_1 < c_1 - r_1$ is a weakly dominated strategy for Firm 1; and (ii) setting $p_2 < c_2 - r_H$ is a weakly dominated strategy for Firm 2.

Lemma A15. Suppose Assumption 1 holds. Then an equilibrium does not exist in which $p_1 \ge p_2 + G_1 - G_2 + K_2$ and $p_1 > c_1 - r_L$.

Lemma A16. Suppose Assumption 1 holds and $G_2 - G_1 + c_1 - c_2 - K_2 \neq 0$. Then an equilibrium does not exist in which $p_1 = p_2 + G_1 - G_2 + K_2$ and $p_1 = c_1 - r_L$.

Lemma A17. Suppose Assumption 1 holds. An equilibrium does not exist in which $p_1 = p_2 + G_1 - G_2 + K_2$ and $p_1 \le c_1 - r_1$.

Lemma A18. Suppose Assumption 1 holds. Then an equilibrium does not exist in which $p_2 \ge p_1 + G_2 - G_1 + K_1$ and $p_2 \ne c_2 - r_H$.

Proof of Lemma 1. The proof follows from the following lemmas (Lemmas A1.1 – A1.6).

Lemma A1.1. Suppose the default privacy setting cannot be changed. Then: (i) all users buy a phone from Firm 1 if $p_2 - p_1 > G_2 - G_1 + t$; and (ii) all users buy a phone from Firm 2 if $p_2 - p_1 < G_2 - G_1 - t$.

<u>Proof.</u> To prove conclusion (i), observe that all users buy a phone from Firm 1 if, for all $x \in [0, 1]$:

$$G_1 - tx - p_1 > G_2 - t[1 - x] - p_2 \iff x < \frac{1}{2} + \frac{1}{2t} [G_1 - G_2 - p_1 + p_2].$$
(6)

(6) holds for all $x \in [0, 1]$ if:

$$1 < \frac{1}{2} + \frac{1}{2t} \left[G_1 - G_2 - p_1 + p_2 \right] \Leftrightarrow p_2 - p_1 > G_2 - G_1 + t$$

The proof of conclusion (ii) is analogous. \Box

Lemma A1.2. Suppose the default privacy setting cannot be changed and t > 3 |A|. Then no equilibrium exists in which one firm serves all users.

<u>Proof</u>. First suppose Firm 1 serves all users. Then Lemma A1.1 implies that for all p_2 that generate nonnegative profit for Firm 2:

$$p_1 \leq p_2 + G_1 - G_2 - t \,. \tag{7}$$

(7) holds for all such p_2 if:

$$p_1 \leq c_2 - r_H + G_1 - G_2 - t.$$
(8)

Firm 1's profit when it serves all users at a price that satisfies (8) is:

$$\pi_{1} = p_{1} + r_{L} - c_{1} \leq c_{2} - r_{H} + G_{1} - G_{2} - t + r_{L} - c_{1}$$

$$= G_{1} + r_{L} - c_{1} - (G_{2} + r_{H} - c_{2}) - t < 0 \text{ when } t > 3 |A|.$$
(9)
27

(9) implies that no equilibrium exists in which Firm 1 serves all users.

The proof for the case where Firm 2 serves all users is analogous. \Box

Lemma A1.3. Suppose the default privacy setting cannot be changed and $p_2 - p_1 \in [G_2 - G_1 - t, G_2 - G_1 + t]$. Then: (i) a user located at $x_0 \equiv \frac{1}{2t} [t + G_1 - G_2 + p_2 - p_1] \in [0, 1]$ is indifferent between purchasing a phone from Firm 1 and from Firm 2; (ii) if $x_0 > 0$, all users located in $[0, x_0)$ buy a phone from Firm 1; and (iii) if $x_0 < 1$, all users located in $(x_0, 1]$ buy a phone from Firm 2.

<u>Proof</u>. A user located at x is indifferent between purchasing a phone from Firm 1 and from Firm 2 if:

$$G_{1} - tx - p_{1} = G_{2} - t[1 - x] - p_{2} \Leftrightarrow t[1 - 2x] = G_{2} - G_{1} - p_{2} + p_{1}$$

$$\Leftrightarrow 1 - 2x = \frac{1}{t}[G_{2} - G_{1} - p_{2} + p_{1}] \Leftrightarrow x = \frac{1}{2t}[t + G_{1} - G_{2} + p_{2} - p_{1}] \equiv x_{0}$$

$$\Rightarrow x_{0} \in [0, 1] \Leftrightarrow p_{2} - p_{1} \in [G_{2} - G_{1} - t, G_{2} - G_{1} + t].$$

If $x_0 > 0$, then a user located at $x \in [0, x_0)$ buys a phone from Firm 1 because:

$$G_1 - tx - p_1 > G_2 - t[1 - x] - p_2 \iff x < \frac{1}{2t} [t + G_1 - G_2 + p_2 - p_1] = x_0.$$

If $x_0 < 1$, then a user located at $x \in (x_0, 1]$ buys a phone from Firm 2 because:

$$G_2 - t[1 - x] - p_2 > G_1 - tx - p_1 \iff x > \frac{1}{2t}[t + G_1 - G_2 + p_2 - p_1] = x_0.$$

Lemma A1.4. Suppose the default privacy setting cannot be changed and t > 3 |A|. Then in equilibrium, there exists a $x_0 \in [0,1]$ such that: (i) a user located at x_0 is indifferent between buying a phone from Firm 1 and from Firm 2; (ii) all users located in $[0, x_0)$ buy a phone from Firm 1; and (iii) all users located in $(x_0, 1]$ buy a phone from Firm 2. Furthermore: $p_1 = c_1 - r_L + t - A$; $p_2 = c_2 - r_H + t + A$; $\pi_1 = \frac{1}{2t} [t - A]^2$; and $\pi_2 = \frac{1}{2t} [t + A]^2$.

<u>Proof.</u> Lemma A1.2 implies that Firm 1 and Firm 2 both serve some users in equilibrium. Therefore, Lemma A1.1 implies that $p_2 - p_1 \in [G_2 - G_1 - t, G_2 - G_1 - t]$. Consequently, Lemma A1.3 implies that a user located at

$$x_0 \equiv \frac{1}{2t} \left[t + G_1 - G_2 + p_2 - p_1 \right] \in [0, 1]$$
(10)

is indifferent between purchasing a phone from Firm 1 and from Firm 2. Furthermore, all users located in $[0, x_0)$ buy a phone from Firm 1, and all users located in $(x_0, 1]$ buy a phone from Firm 2. Therefore, (10) implies that Firm 1's profit is:

$$\pi_1 = \left[p_1 + r_L - c_1 \right] \frac{1}{2t} \left[t + G_1 - G_2 + p_2 - p_1 \right].$$
(11)

The unique value of p_1 that maximizes π_1 in (11) is given by:

$$\frac{\partial \pi_1}{\partial p_1} = 0 \iff p_1 = \frac{1}{2} \left[t + c_1 - r_L + G_1 - G_2 + p_2 \right].$$
(12)

(10) and Lemma A1.3 imply that Firm 2's profit is:

$$\pi_2 = [p_2 + r_H - c_2] \frac{1}{2t} [t + G_2 - G_1 + p_1 - p_2].$$
(13)

The unique value of p_2 that maximizes π_2 in (13) is given by:

$$\frac{\partial \pi_2}{\partial p_2} = 0 \quad \Leftrightarrow \quad p_2 = \frac{1}{2} \left[t + c_2 - r_H + G_2 - G_1 + p_1 \right]. \tag{14}$$

(12) and (14) imply:

$$p_{1} = \frac{1}{2} \left[t + c_{1} - r_{L} + G_{1} - G_{2} \right] + \frac{1}{4} \left[t + c_{2} - r_{H} + G_{2} - G_{1} + p_{1} \right]$$

$$\Rightarrow p_{1} = \frac{1}{3} \left[3t - 3A + 3(c_{1} - r_{L}) \right] = c_{1} - r_{L} + t - A.$$
(15)

(14) and (15) imply:

$$p_2 = \frac{1}{3} \left[3t + 3A + 3(c_2 - r_H) \right] = c_2 - r_H + t + A.$$
 (16)

(15) implies that Firm 1's profit margin is positive because:

$$p_1 + r_L - c_1 = t - A > 0. (17)$$

(16) implies that Firm 2's profit margin is positive because:

$$p_2 + r_H - c_2 = t + A > 0. (18)$$

(15) and (16) imply:

$$p_2 - p_1 = \frac{1}{3} \left[c_2 - c_1 + 2 G_2 - 2 G_1 + r_L - r_H \right].$$
(19)

(15), (19), and Lemma A1.3 imply:

$$\pi_1 = \frac{1}{3} \left[3t + 2c_1 + c_2 + G_1 - G_2 - 2r_L - r_H + 3r_L - 3c_1 \right]$$

$$\cdot \frac{1}{2t} \left[t + G_1 - G_2 + \frac{1}{3} \left(c_2 - c_1 + 2G_2 - 2G_1 + r_L - r_H \right) \right] = \frac{1}{2t} \left[t - A \right]^2.$$

(16), (19), and Lemma A1.3 imply:

$$\pi_2 = \frac{1}{3} \left[3t + 2c_2 + c_1 + G_2 - G_1 - 2r_H - r_L + 3r_H - 3c_2 \right] \\ \cdot \frac{1}{2t} \left[t + G_2 - G_1 + \frac{1}{3} \left(c_1 - c_2 + 2G_1 - 2G_2 + r_H - r_L \right) \right] = \frac{1}{2t} \left[t + A \right]^2. \ \Box$$

Lemma A1.5. Suppose the default privacy setting cannot be changed, A < 0, and $t \in (|A|, 3|A|)$. Then at the unique equilibrium, both firms sell phones, Firm 1's profit is $\pi_1 = \frac{1}{2t} [t - A]^2 > 0$ and Firm 2's profit is $\pi_2 = \frac{1}{2t} [t + A]^2 > 0$.

<u>Proof</u>. First suppose that an equilibrium exists in which all users buy a phone from Firm 2. Then because the user located at 0 buys a phone from Firm 2:

$$G_2 - p_2 - t > G_1 - p_1 \iff p_2 < p_1 + G_2 - G_1 - t.$$
 (20)

(20) must hold for all p_1 for which Firm 1's profit margin is positive. Therefore:

$$p_2 < c_1 - r_L + G_2 - G_1 - t \equiv \hat{p}_2.$$
 (21)

Firm 2's profit when it sets a price marginally below \hat{p}_2 is nearly:

$$\pi_2 = \widehat{p}_2 + r_H - c_2 = c_1 - r_L + G_2 - G_1 - t + r_H - c_2$$

= $G_2 + r_H - c_2 - (G_1 + r_L - c_1) - t = 3A - t < 0.$ (22)

(22) implies that an equilibrium in which all users buy a phone from Firm 2 does not exist under the specified conditions.

Now suppose that an equilibrium exists in which all users buy a phone from Firm 1. Then because the user located at 1 buys a phone from Firm 1:

$$G_1 - p_1 - t > G_2 - p_2 \iff p_1 < p_2 + G_1 - G_2 - t.$$
 (23)

(23) must hold for all p_2 for which Firm 2's profit margin is positive. Therefore:

$$p_1 < c_2 - r_H + G_1 - G_2 - t \equiv \hat{p}_1.$$
 (24)

Firm 1's profit when it sets a price marginally below \hat{p}_1 is nearly:

$$\pi_{1} = \hat{p}_{1} + r_{L} - c_{1} = c_{2} - r_{H} + G_{1} - G_{2} - t + r_{L} - c_{1}$$

= $G_{1} + r_{L} - c_{1} - (G_{2} + r_{H} - c_{2}) - t = -3A - t > 0.$ (25)

If a user located at $x \in [0, 1]$ is indifferent between purchasing a phone from Firm 1 and

from Firm 2, then:

$$G_1 - tx - p_1 = G_2 - t[1 - x] - p_2 \iff x = \frac{1}{2t} [t + G_1 - G_2 + p_2 - p_1].$$
(26)

(26) implies that when $p_2 = c_2 - r_H$ and $p_1 \in (\hat{p}_1, \hat{p}_1 + 2t)$, users located in $[0, \hat{x}_0)$ purchase a phone from Firm 1 and users located in $(\hat{x}_0, 1]$ purchase a phone from Firm 2, where:

$$\widehat{x}_0 = \frac{1}{2t} \left[t + G_1 - G_2 + c_2 - r_H - p_1 \right] \in (0, 1).$$

Firm 1's corresponding profit is:

$$\pi_1(p_1) = [p_1 + r_L - c_1] \frac{1}{2t} [t + G_1 - G_2 + c_2 - r_H - p_1].$$
(27)

Differentiating (27) provides:

$$\pi'_{1}(p_{1}) = \frac{1}{2t} \left[t + G_{1} - G_{2} + c_{2} - r_{H} - r_{L} + c_{1} - 2p_{1} \right] \Rightarrow \pi''_{1}(p_{1}) = -\frac{1}{t} < 0.$$
(28)

(24) and (28) imply:

$$\pi'_{1}(p_{1})|_{p_{1}=\widehat{p}_{1}} = \frac{1}{2t} \left[3t + 3A \right] = \frac{3}{2t} \left[t + A \right] > 0.$$
⁽²⁹⁾

(28) and (29) imply that when $p_2 = c_2 - r_H$, Firm 1 will increase p_1 above \hat{p}_1 , thereby ensuring that both firms sell phones. Consequently, the analysis in the proof of Lemma A1.4 implies that at the unique equilibrium, both firms sell phones, Firm 1's profit is $\pi_1 = \frac{1}{2t} [t - A]^2 > 0$ and Firm 2's profit is $\pi_2 = \frac{1}{2t} [t + A]^2 > 0$. \Box

Lemma A1.6. Suppose the default privacy setting cannot be changed, A > 0, and $t \in (|A|, 3|A|)$. Then at the unique equilibrium, both firms sell phones, Firm 1's profit is $\pi_1 = \frac{1}{2t} [t - A]^2 > 0$ and Firm 2's profit is $\pi_2 = \frac{1}{2t} [t + A]^2 > 0$.

<u>Proof.</u> The proof is analogous to the proof of Lemma A1.5. $\Box \blacksquare$

Proof of Lemma 2. The proof follows directly from the following lemmas (Lemmas A2.1 – A2.4).

Lemma A2.1. Suppose $K_1 = K_2 = 0$. Then: (i) a user located in $[0, \frac{1}{2})$ will change the default setting on the phone she purchases if and only if she purchases the phone from Firm 2; (ii) a user located in $(\frac{1}{2}, 1]$ will change the default setting on the phone she purchases if and only if she purchases the phone from Firm 1; and (iii) a user located at $\frac{1}{2}$ will not change the default setting on the phone she purchases.

<u>Proof</u>. The conclusions follow directly from the proofs of Lemmas A1 – A3. \Box

Lemma A2.2. Suppose $K_1 = K_2 = 0$. Then: (i) all users buy a phone from Firm 1 if $p_2 > p_1 + G_2 - G_1$; (ii) all users buy a phone from Firm 2 if $p_2 < p_1 + G_2 - G_1$; and (iii) all users are indifferent between buying a phone from Firm 1 and from Firm 2 if $p_2 = p_1 + G_2 - G_1$.

<u>Proof.</u> Lemma A2.1 implies that a user located at $x_1 \in [0, \frac{1}{2})$ will buy a phone from Firm 1 if:

$$G_1 - t x_1 - p_1 > G_2 - t x_1 - p_2 \iff p_2 > p_1 + G_2 - G_1$$

Lemma A2.1 also implies that a user located at $x_2 \in (\frac{1}{2}, 1]$ will buy a phone from Firm 1 if:

$$G_1 - t[1 - x_2] - p_1 > G_2 - t[1 - x_2] - p_2 \iff p_2 > p_1 + G_2 - G_1.$$

Lemma A2.1 further implies that a user located at $\frac{1}{2}$ will buy a phone from Firm 1 if:

$$G_1 - \frac{1}{2}t - p_1 > G_2 - \frac{1}{2}t - p_2 \Leftrightarrow p_2 > p_1 + G_2 - G_1.$$

The proofs of the remaining conclusions are analogous, and so are omitted. \Box

Lemma A2.3. Suppose $K_1 = K_2 = 0$ and $G_2 - c_2 > G_1 - c_1$. Then in equilibrium, all users purchase a phone from Firm 2 at a price just below $c_1 - \frac{1}{2} [r_H + r_L] + G_2 - G_1$. Firm 1's profit is 0. Firm 2's profit is (nearly) $G_2 - c_2 - (G_1 - c_1)$.

<u>Proof</u>. Lemmas A2.1 and A2.2 imply that for $\varepsilon_1 > 0$, Firm 2's expected profit is:

$$\pi_{2} = \begin{cases} 0 & \text{if } p_{2} > p_{1} + G_{2} - G_{1} \\ \frac{1}{2} \left[p_{1} + \frac{r_{L} + r_{H}}{2} + G_{2} - G_{1} - c_{2} \right] & \text{if } p_{2} = p_{1} + G_{2} - G_{1} \\ p_{1} + \frac{r_{L} + r_{H}}{2} + G_{2} - G_{1} - c_{2} - \varepsilon_{1} & \text{if } p_{2} = p_{1} + G_{2} - G_{1} - \varepsilon_{1} . \end{cases}$$
(30)

Firm 1 must secure nonnegative profit in equilibrium. Therefore, in any equilibrium in which all users either strictly prefer to purchase a phone from Firm 1 or are indifferent between purchasing a phone from Firm 1 and Firm 2, it must be the case that $p_1 \ge c_1 - \frac{1}{2} [r_H + r_L]$. Consequently, in any such equilibrium:

$$p_1 + \frac{r_L + r_H}{2} + G_2 - G_1 - c_2 \ge G_2 - c_2 - (G_1 - c_1) > 0.$$
(31)

(30) and (31) imply that for ε_1 sufficiently small, Firm 2 secures strictly higher profit by setting $p_2 = p_1 + G_2 - G_1 - \varepsilon_1$ than by setting $p_2 \ge p_1 + G_2 - G_1$. Therefore, in equilibrium, Firm 2 will set p_2 just below $c_1 - \frac{1}{2} [r_H + r_L] + G_2 - G_1$ to ensure that Firm 1 cannot profitably attract any users. Consequently, Firm 1's profit is 0 and Firm 2's profit is nearly:

$$c_1 - \frac{1}{2} [r_H + r_L] + G_2 - G_1 + \frac{1}{2} [r_H + r_L] - c_2 = G_2 - c_2 - (G_1 - c_1).$$

Lemma A2.4. Suppose $K_1 = K_2 = 0$ and $G_1 - c_1 > G_2 - c_2$. Then in equilibrium, all users purchase a phone from Firm 1 at a price just below $p_1 = c_2 - \frac{1}{2} [r_H + r_L] + G_1 - G_2$. Firm 2's profit is 0. Firm 1's profit is (nearly) $G_1 - c_1 - (G_2 - c_2)$.

<u>Proof.</u> The proof is analogous to the proof of Lemma A2.3. $\Box \blacksquare$

<u>Proof of Proposition 1</u>. (10) and Lemmas A2 – A4 and A11 imply that in any equilibrium with the identified properties, Firm 1's profit is:

$$\pi_1 = [p_1 + r_L - c_1] \frac{1}{2t} [t + G_1 - G_2 + p_2 - p_1].$$
(32)

The unique value of p_1 that maximizes π_1 in (32) is given by:

$$\frac{\partial \pi_1}{\partial p_1} = 0 \quad \Leftrightarrow \quad p_1 = \frac{1}{2} \left[t + c_1 - r_L + G_1 - G_2 + p_2 \right]. \tag{33}$$

(10) and Lemmas A1, A3, A4, and A11 imply that in any equilibrium with the identified properties, Firm 2's profit is:

$$\pi_2 = [p_2 + r_H - c_2] \frac{1}{2t} [t + G_2 - G_1 + p_1 - p_2].$$
(34)

The unique value of p_2 that maximizes π_2 in (34) is given by:

$$\frac{\partial \pi_2}{\partial p_2} = 0 \iff p_2 = \frac{1}{2} \left[t + c_2 - r_H + G_2 - G_1 + p_1 \right].$$
(35)

(33) and (35) imply that in any equilibrium:

$$p_{1} = \frac{1}{2} \left[t + c_{1} - r_{L} + G_{1} - G_{2} \right] + \frac{1}{4} \left[t + c_{2} - r_{H} + G_{2} - G_{1} + p_{1} \right]$$

$$\Rightarrow p_{1} = \frac{1}{3} \left[3t - 3A + 3(c_{1} - r_{L}) \right] = c_{1} - r_{L} + t - A.$$
(36)

(35) and (36) imply:

$$p_{2} = \frac{1}{2} \left[t + c_{2} - r_{H} + G_{2} - G_{1} \right] + \frac{1}{6} \left[3t + 2c_{1} + c_{2} + G_{1} - G_{2} - 2r_{L} - r_{H} \right]$$
$$= \frac{1}{3} \left[3t + 3A + 3(c_{2} - r_{H}) \right] = c_{2} - r_{H} + t + A.$$
(37)

(36) and the maintained assumptions imply that Firm 1's profit margin is positive because:

$$p_1 + r_L - c_1 = t - A > 0.$$

(37) and the maintained assumptions imply that Firm 2's profit margin is positive because: $p_2 + r_H - c_2 = t + A > 0.$

(36) and Condition 1A imply that $p_1 \ge 0$ because:

$$p_{1} \geq 0 \iff 3t + 2c_{1} + c_{2} + G_{1} - G_{2} - 2r_{L} - r_{H} \geq 0$$

$$\Leftrightarrow A = \frac{1}{3} [G_{2} + r_{H} - c_{2} - (G_{1} + r_{L} - c_{1})] \leq t - r_{L} + c_{1}.$$
(38)

(37) and Condition 1A imply that $p_2 \ge 0$ because:

$$p_{2} \geq 0 \iff 3t + 2c_{2} + c_{1} + G_{2} - G_{1} - 2r_{H} - r_{L} \geq 0$$

$$\Leftrightarrow A = \frac{1}{3} [G_{2} + r_{H} - c_{2} - (G_{1} + r_{L} - c_{1})] \geq r_{H} - c_{2} - t.$$
(39)

(36) and (37) imply:

$$p_2 - p_1 = \frac{1}{3} \left[c_2 - c_1 + 2G_2 - 2G_1 + r_L - r_H \right].$$
(40)

(10) and (40) imply that the user who is indifferent between purchasing a phone from Firm 1 and Firm 2 is located at:

$$x_0 = \frac{1}{2} - \frac{1}{6t} \left[G_2 + r_H - c_2 - (G_1 + r_L - c_1) \right] = \frac{1}{2} - \frac{A}{2t}.$$
 (41)

(41) and the maintained assumptions imply that $x_0 \in (\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2} + \frac{K_1}{2t})$, so no user changes the default setting on the phone she purchases (from Lemmas A1 – A4).

(41) and (36) imply:

$$\pi_1 = [p_1 + r_L - c_1] x_0 = [t - A] \left[\frac{t - A}{2t} \right] = \frac{1}{2t} [t - A]^2.$$
(42)

(41) and (37) imply:

$$\pi_2 = [p_2 + r_H - c_2] [1 - x_0] = [t + A] \left[\frac{t + A}{2t} \right] = \frac{1}{2t} [t + A]^2.$$
(43)

(40) implies:

$$p_1 - p_2 > G_1 - G_2 - K_1 \iff K_1 > \frac{1}{3} [G_1 - G_2 + c_2 - c_1 + r_L - r_H] = -A.$$
 (44)

(40) also implies:

$$p_1 - p_2 < G_1 - G_2 + K_2 \iff K_2 > \frac{1}{3} [G_2 - G_1 + c_1 - c_2 + r_H - r_L] = A.$$
 (45)

(44), (45), and the maintained assumptions imply:

$$p_1 - p_2 \in (G_1 - G_2 - K_1, G_1 - G_2 + K_2).$$
 (46)

The foregoing analysis and Lemma A11 imply that the identified putative equilibrium is unique among equilibria in which (46) holds. It remains to verify that neither firm can increase its profit by unilaterally changing its price so that (46) does not hold. We first show this is the case for Firm 1.

Lemmas A7 and A8 imply that if Firm 1 sets $p_1 > p_2 + G_1 - G_2 + K_2$, then no users purchase a phone from Firm 1. Therefore, Firm 1's profit (0) is less than the profit specified in (42).

If Firm 1 sets $p_1 = p_2 + G_1 - G_2 + K_2$ when p_2 is as specified in (37), then:

$$\begin{array}{rcl} p_1 &=& p_2 + G_1 - G_2 + K_2 &=& c_2 - r_H + t + A + G_1 - G_2 + K_2 \\ \\ &=& -3\,A + c_1 - r_L + t + A + K_2 &=& c_1 - r_L - 2\,A + t + K_2 \\ \\ &>& c_1 - r_L \ \Leftrightarrow \ t + K_2 - 2\,A \ > \ 0 \,. \end{array}$$

The last inequality holds here because t > A and $K_2 > A$, by assumption. Because $p_1 = p_2 + G_1 - G_2 + K_2$ and $p_1 > c_1 - r_L$, the proof of Lemma A15 implies that Firm 1 can increase its profit when p_2 is as specified in (37) by choosing p_1 to ensure $p_1 - p_2 \in (G_1 - G_2 - K_1, G_1 - G_2 + K_2)$. Therefore, Firm 1 cannot increase its profit above the profit specified in (42) by setting $p_1 = p_2 + G_1 - G_2 + K_2$ when p_2 is as specified in (37).

Lemmas A9 and A10 imply that if Firm 1 sets $p_1 < p_2 + G_1 - G_2 - K_1$, then all users purchase a phone from Firm 1. (4) implies that the maximum profit Firm 1 can secure by setting such a price when p_2 is as specified in (37) is nearly:

$$\pi_{1D} = p_2 + G_1 - G_2 - K_1 + r_1 - c_1 = c_2 - r_H + t + A + G_1 - G_2 - K_1 + r_1 - c_1$$

= $G_1 - c_1 - (G_2 + r_H - c_2) + A + t + r_1 - K_1$
= $-2A + t - K_1 + \frac{r_H - r_L}{2t} [t - K_1] = -2A + \frac{t - K_1}{2t} [2t + r_H - r_L].$ (47)

(42) and (47) imply that Firm 1 cannot increase its profit by setting $p_1 < p_2 + G_1 - G_2 - K_1$ when p_2 is as specified in (37) if Condition 1B holds.

If Firm 1 sets $p_1 = p_2 + G_1 - G_2 - K_1$ when p_2 is as specified in (37), then Lemma A6 implies that: (i) all users located in $\left[\frac{1}{2} + \frac{K_1}{2t}, 1\right]$ are indifferent between buying a phone from Firm 1 and from Firm 2; and (ii) all users located in $\left[0, \frac{1}{2} + \frac{K_1}{2t}\right]$ buy a phone from Firm 1. Therefore, Lemma A2 implies that Firm 1's profit is:

$$\pi_{1} = \left[p_{1} + r_{L} - c_{1}\right] \left[\frac{1}{2} + \frac{K_{1}}{2t}\right] + \left[p_{1} + r_{H} - c_{1}\right] \frac{1}{2} \left[\frac{1}{2} - \frac{K_{1}}{2t}\right].$$

$$< \left[p_{1} + r_{L} - c_{1}\right] \left[\frac{1}{2} + \frac{K_{1}}{2t}\right] + \left[p_{1} + r_{H} - c_{1}\right] \left[\frac{1}{2} - \frac{K_{1}}{2t}\right]$$

$$(48)$$

$$= p_1 + r_1 - c_1 = \pi_{1D} < \frac{1}{2t} [t - A]^2.$$
(49)

The first inequality in (49) holds because $p_1 + r_H - c_1$ must be strictly positive if Firm 1 is to secure positive profit in this case. The last inequality in (49) reflects (47) and Condition 1B. (42) and (49) imply that Firm 1 will not set $p_1 = p_2 + G_1 - G_2 - K_1$ when p_2 is as specified in (37).

Now we show that Firm 2 cannot increase its profit by unilaterally changing its price so that (46) does not hold when p_1 is as specified in (36).

Lemmas A9 and A10 imply that if Firm 2 sets $p_2 > p_1 + G_2 - G_1 + K_1$, then no users purchase a phone from Firm 2, so Firm 2's profit (0) is less than the profit specified in (43).

If Firm 2 sets $p_2 = p_1 + G_2 - G_1 + K_1$ when p_1 is as specified in (36), then:

$$p_{2} = p_{1} + G_{2} - G_{1} + K_{1} = c_{1} - r_{L} + t - A + G_{2} - G_{1} + K_{1}$$

$$= G_{2} + r_{H} - c_{2} - G_{1} - r_{L} + c_{2} + t - A + K_{1} + c_{2} - r_{H}$$

$$= 3A + t - A + K_{1} + c_{2} - r_{H} = 2A + t + K_{1} + c_{2} - r_{H} > c_{2} - r_{H}.$$

The last inequality holds here because $K_1 > -A$ and t > -A, by assumption. Because $p_2 = p_1 + G_2 - G_1 + K_1 > c_2 - r_H$ when p_1 is as specified in (36), the proof of Lemma A20 implies that Firm 2 can increase its profit by setting p_2 to ensure $p_1 - p_2 \in (G_1 - G_2 - K_1, G_1 - G_2 + K_2)$. Therefore, Firm 2 cannot increase its profit by setting $p_2 = p_1 + G_2 - G_1 + K_1$ when p_1 is as specified in (36).

Lemmas A7 and A8 imply that if Firm 2 sets $p_2 < p_1 + G_2 - G_1 - K_2$, then all users purchase a phone from Firm 2. (4) implies that the maximum profit Firm 2 can secure by setting such a price when p_1 is as specified in (36) is nearly:

$$\pi_{2D} = p_1 + G_2 - G_1 - K_2 + r_2 - c_2 = c_1 - r_L + t - A + G_2 - G_1 - K_2 + r_2 - c_2$$

= $G_2 - c_2 - (G_1 + r_L - c_1) - A + t + r_2 - K_2$
= $2A + [t - K_2] \left[1 - \frac{1}{2t} (r_H - r_L) \right] = 2A + \frac{t - K_2}{2t} [2t - r_H + r_L].$ (50)

(43) and (50) imply that Firm 2 cannot increase its profit by setting $p_2 < p_1 + G_2 - G_1 - K_2$ when p_1 is as specified in (36) if Condition 1B holds.

If Firm 2 sets $p_2 = p_1 + G_2 - G_1 - K_2$ when p_1 is as specified in (36), Lemma A5 implies that: (i) all users located in $[0, \frac{1}{2} - \frac{K_2}{2t}]$ are indifferent between buying a phone from Firm 1 and from Firm 2; and (ii) all users located in $(\frac{1}{2} - \frac{K_2}{2t}, 1]$ buy a phone from Firm 2. Therefore, Lemmas A1, A3, and A5 imply that Firm 2's profit is:

$$\pi_2 = \left[p_2 + r_H - c_2 \right] \left[\frac{1}{2} + \frac{K_2}{2t} \right] + \left[p_2 + r_L - c_2 \right] \frac{1}{2} \left[\frac{1}{2} - \frac{K_2}{2t} \right].$$
(51)

If $p_2 = p_1 + G_2 - G_1 - K_2$ and p_1 is as specified in (36):

$$p_2 + r_H - c_2 = 3A + t - A - K_2 = 2A + t - K_2$$
(52)

$$\Rightarrow p_2 + r_L - c_2 = p_2 + r_H - c_2 - r_H + r_L = 2A + t - K_2 - r_H + r_L.$$
(53)

(52) implies that if Firm 2 is to secure positive profit under the presumed deviation, it must be the case that $2A + t - K_2 > 0$.

Initially suppose $p_2 + r_L - c_2 = 2A + t - K_2 - r_H + r_L > 0$. Then (51) – (53) imply:

$$\pi_{2} = \left[p_{2} + r_{H} - c_{2}\right] \left[\frac{1}{2} + \frac{K_{2}}{2t}\right] + \left[p_{2} + r_{L} - c_{2}\right] \frac{1}{2} \left[\frac{1}{2} - \frac{K_{2}}{2t}\right]$$

$$< \left[p_{2} + r_{H} - c_{2}\right] \left[\frac{1}{2} + \frac{K_{2}}{2t}\right] + \left[p_{2} + r_{L} - c_{2}\right] \left[\frac{1}{2} - \frac{K_{2}}{2t}\right]$$

$$= 2A + t - K_{2} - \frac{1}{2t} \left[r_{H} - r_{L}\right] \left[t - K_{2}\right] = \pi_{2D} < \frac{1}{2t} \left[t + A\right]^{2}.$$
(54)

(54) implies that Firm 2 cannot increase its profit by undertaking the proposed deviation in this case.

Now suppose $p_2 + r_L - c_2 = 2A + t - K_2 - r_H + r_L \leq 0$. (51) implies that Firm 2's profit is maximized in this case when $p_2 + r_L - c_2 = 2A + t - K_2 - r_H + r_L = 0$. (51) and (52) imply that Firm 2's maximum profit in this case is:

$$\pi_2 = \left[p_2 + r_H - c_2 \right] \left[\frac{1}{2} + \frac{K_2}{2t} \right] = \frac{1}{2t} \left[2A + t - K_2 \right] \left[t + K_2 \right].$$

Observe that:

$$\frac{1}{2t} [t+A]^2 > \frac{1}{2t} [2A+t-K_2] [t+K_2] \iff [A-K_2]^2 > 0.$$
 (55)

(43) and (55) imply that Firm 2 cannot increase its profit by undertaking the proposed deviation in this case. \blacksquare

Proof of Proposition 2. We first show that the identified p_1 maximizes Firm 1's profit when $p_2 = c_2 - r_H$.

Lemmas A9 and A10 imply that if $p_2 = c_2 - r_H$, then among all $p_1 \leq p_2 + G_1 - G_2 - K_1$, the profit-maximizing p_1 for Firm 1 is marginally below:

$$c_2 - r_H + G_1 - G_2 - K_1 = c_1 - r_L - 3A - K_1 > 0.$$

The inequality here reflects Condition 2D. Lemma A2 implies that Firm 1's corresponding profit is approximately:

$$\pi_{1} = \left[p_{1} + r_{L} - c_{1}\right] \left[\frac{1}{2} + \frac{K_{1}}{2t}\right] + \left[p_{1} + r_{H} - c_{1}\right] \left[\frac{1}{2} - \frac{K_{1}}{2t}\right]$$

$$= G_{1} - G_{2} + c_{2} - c_{1} - \frac{1}{2} \left[r_{H} - r_{L}\right] - K_{1} \left[\frac{2t + r_{H} - r_{L}}{2t}\right]$$

$$= \left[\frac{t - K_{1}}{2t}\right] \left[r_{H} - r_{L}\right] - 3A - K_{1}.$$
(56)

The expression in (56) is strictly positive because $K_1 \leq \frac{2t}{2t+r_H-r_L} \Omega_1$ (from Condition 2C), and because Condition 2B implies:

$$G_1 - G_2 + c_2 - c_1 - \frac{1}{2} \left[r_H - r_L \right] = \frac{1}{2} \left[r_H - r_L \right] - 3A \ge \Omega_1 > 0.$$

We now show that Firm 1 cannot increase its profit by setting $p_1 \in (p_2 + G_1 - G_2 - K_1, p_2 + G_1 - G_2 + K_2)$ or $p_1 \ge p_2 + G_1 - G_2 + K_2$ when $p_2 = c_2 - r_H$.

(33) implies that when $p_1 \in (p_2 + G_1 - G_2 - K_1, p_2 + G_1 - G_2 + K_2)$, the price that maximizes Firm 1's profit when $p_2 = c_2 - r_H$ is:

$$p_1 = \frac{1}{2} \left[t + c_1 - r_L + G_1 - G_2 + p_2 \right] = \frac{1}{2} \left[t + G_1 - G_2 + c_1 + c_2 - r_H - r_L \right].$$
(57)

(32) and (57) Firm 1's corresponding profit is:

$$\pi_{1}^{'} = \frac{1}{8t} \left[t + G_{1} - G_{2} + c_{2} - c_{1} - r_{H} + r_{L} \right]^{2}.$$
(58)

(56) and (58) imply that Firm 1 cannot increase its profit by setting $p_1 \in (p_2 + G_1 - G_2 - K_1, p_2 + G_1 - G_2 + K_2)$ because:

$$\pi_1 \geq \pi'_1 \iff K_1 \leq \frac{2t}{2t + r_H - r_L} \Omega_1.$$
(59)

Lemmas A7 and A8 imply that if Firm 1 sets price $p_1 > p_2 + G_1 - G_2 + K_2$, it will sell no phones, and so will make 0 profit. Therefore, among all $p_1 \ge p_2 + G_1 - G_2 + K_2$, the price that maximizes Firm 1's profit is $p_1 = p_2 + G_1 - G_2 + K_2$. When $p_2 = c_2 - r_H$, this price is $p_1 = c_2 - r_H + G_1 - G_2 + K_2$. Lemma A5 implies that when $p_1 = p_2 + G_1 - G_2 + K_2$: (i) all users located in $[0, \frac{1}{2} - \frac{K_2}{2t}]$ are indifferent between buying a phone from Firm 1 and from Firm 2; and (ii) all users located in $(\frac{1}{2} - \frac{K_2}{2t}, 1]$ buy a phone from Firm 2. Therefore, Lemma A2 implies that Firm 1's profit is:

$$\widetilde{\widetilde{\pi}}_{1} = \frac{1}{2} \left[G_{1} - G_{2} - c_{1} + c_{2} - r_{H} + r_{L} + K_{2} \right] \left[\frac{1}{2} - \frac{K_{2}}{2t} \right].$$
(60)

If $G_1 - G_2 - c_1 + c_2 - r_H + r_L + K_2 \leq 0$, then $\widetilde{\widetilde{\pi}}_1 \leq 0$. Therefore, Firm 1 will never set

 $p_1 \ge p_2 + G_1 - G_2 + K_2$ when $p_2 = c_2 - r_H$ in this case.

If $G_1 - G_2 - c_1 + c_2 - r_H + r_L + K_2 > 0$, then $\tilde{\pi}_1 > 0$. In this case, if Firm 1 were to reduce its price to $p_1 = c_2 - r_H + G_1 - G_2 + K_2 - \varepsilon_2$ where $\varepsilon_2 > 0$, all users located in $[0, \frac{1}{2} - \frac{K_2}{2t}]$ would purchase a phone from Firm 1. Consequently, Firm 1's profit would be at least:

$$\pi_{1} = \left[p_{1} - \varepsilon_{2} + r_{L} - c_{1}\right] \left[\frac{1}{2} - \frac{K_{2}}{2t}\right]$$
$$= \widetilde{\widetilde{\pi}}_{1} + \widetilde{\widetilde{\pi}}_{1} - \varepsilon_{2} \left[\frac{1}{2} - \frac{K_{2}}{2t}\right] > \widetilde{\widetilde{\pi}}_{1} \quad \text{for sufficiently small } \varepsilon_{2}.$$
(61)

(61) implies that Firm 1 could increase its profit by reducing its price marginally below $p_2+G_1-G_2+K_2$ when $p_2 = c_2 - r_H$. Therefore, Firm 1 will never set $p_1 \ge p_2+G_1-G_2+K_2$ when $p_2 = c_2 - r_H$.

In summary, we have established that when $p_2 = c_2 - r_H$, Firm 1 maximizes its profit by setting p_1 marginally below $c_2 - r_H + G_1 - G_2 - K_1$.

We now show that when Firm 1 sets p_1 marginally below $c_2 - r_H + G_1 - G_2 - K_1$, Firm 2 maximizes its profit by setting $p_2 = c_2 - r_H$. Lemmas A9 and A10 imply that when Firm 1 sets p_1 marginally below $c_2 - r_H + G_1 - G_2 - K_1$, Firm 2 attracts no users (and so secures no profit) if it sets $p_2 = c_2 - r_H$. Firm 2 continues to attract no users (and so continues to secure no profit) if it sets $p_2 > c_2 - r_H$. Firm 2 incurs negative profit if it sets $p_2 < c_2 - r_H$. Therefore, Firm 2 cannot increase its profit by setting $p_2 \neq c_2 - r_H$ when Firm 1 sets p_1 marginally below $c_2 - r_H + G_1 - G_2 - K_1$.

Finally, Lemma A2 implies that all users located in the interval $\left[\frac{1}{2} + \frac{K_1}{2t}, 1\right]$ (and only these users) change the default setting on the phone they purchase from Firm 1.

Proof of Proposition 3. Lemmas A7 and A8 imply that if p_2 is marginally below $p_1 + G_2 - G_1 - K_2$, then all users buy a phone from Firm 2. Therefore, Firm 1's profit is 0.

Among all values of p_2 below $p_1 + G_2 - G_1 - K_2$, the value of p_2 that is most profitable for Firm 2 is marginally below:

$$p_2 = p_1 + G_2 - G_1 - K_2. (62)$$

We first show that if Firm 2 sets p_2 marginally below $p_1+G_2-G_1-K_2$, then Firm 1 will set $p_1 \in [c_1 - r_1, \min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$ in equilibrium. We do so by first explaining why it cannot be the case that $p_1 < c_1 - r_1$ or $p_1 > \min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}$ in equilibrium. Then we explain why, when $p_1 \in [c_1 - r_1, \min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$ and Firm 2 sets p_2 is marginally below $p_1 + G_2 - G_1 - K_2$, Firm 1 cannot strictly increase its profit by setting a different price.

Lemma A14 implies that setting $p_1 < c_1 - r_1$ is a dominated strategy for Firm 1.

Consider a putative equilibrium in which $p_1 > c_1 - r_1 + K_1 + K_2$ and Firm 2 sets p_2 marginally below the p_2 in (62). (62) implies:

$$p_2 > c_1 - r_1 + K_1 + K_2 + G_2 - G_1 - K_2 = G_2 - G_1 + K_1 + c_1 - r_1.$$
 (63)

Lemmas A9 and A10 imply that if Firm 1 sets p_1 marginally below $p_2 + G_1 - G_2 - K_1$, all users will purchase a phone from Firm 1. Consequently, (62) and (63) imply that Firm 1's profit will be nearly:

$$\pi_1 = p_1 + r_1 - c_1 = p_2 + G_1 - G_2 - K_1 + r_1 - c_1$$

> $G_2 - G_1 + K_1 + c_1 - r_1 + G_1 - G_2 - K_1 + r_1 - c_1 = 0$

Because Firm 1 can secure strictly positive profit by deviating from its strategy in the putative equilibrium, the putative equilibrium cannot be an equilibrium.

Next, consider a putative equilibrium in which $p_1 > c_1 - r_L$ and Firm 2 sets p_2 marginally below the p_2 in (62). (62) implies:

$$p_2 > c_1 - r_L + G_2 - G_1 - K_2.$$
 (64)

Lemmas A7 and A8 imply all users buy a phone from Firm 2 when (64) holds. Therefore, Firm 1's profit is 0.

Suppose Firm 1 reduces its price to:

$$p'_1 = p_2 + G_1 - G_2 + K_2 > c_1 - r_L + G_2 - G_1 - K_2 + G_1 - G_2 + K_2 = c_1 - r_L.$$

Lemma A5 implies that when Firm 1 sets price p'_1 : (i) all users located in $[0, \frac{1}{2} - \frac{K_2}{2t}]$ are indifferent between buying a phone from Firm 1 and from Firm 2; and (ii) all users located in $(\frac{1}{2} - \frac{K_2}{2t}, 1]$ buy a phone from Firm 2. Therefore, Lemma A2 implies that Firm 1's profit is:

$$\pi'_{1} = \frac{1}{2} \left[p'_{1} + r_{L} - c_{1} \right] \left[\frac{1}{2} - \frac{K_{2}}{2t} \right] > 0 \text{ (because } p'_{1} > c_{1} - r_{L} \text{)}.$$

Because Firm 1 can secure strictly positive profit by deviating from its strategy in the putative equilibrium, the putative equilibrium cannot be an equilibrium.

We now show that when $p_1 \in [c_1 - r_1, c_1 - r_L]$ and Firm 2 sets p_2 marginally below the p_2 in (62), Firm 1 cannot increase its profit by setting a price $p'_1 \in (p_2 + G_1 - G_2 - K_1, p_2 + G_1 - G_2 + K_2)$. Lemma A11 implies that when Firm 1 sets such a price: (i) all users located in $[0, \frac{1}{2} - \frac{K_2}{2t})$ strictly prefer to buy a phone from Firm 1 than from Firm 2; (ii) all users located in $(\frac{1}{2} + \frac{K_1}{2t}, 1]$ strictly prefer to buy a phone from Firm 2 than from Firm 1; and (iii) some user located in $[\frac{1}{2} - \frac{K_2}{2t}, \frac{1}{2} + \frac{K_1}{2t}]$ is indifferent between buying a phone from Firm 1 and from Firm 2. Furthermore, Lemmas A3 and A4 imply that Firm 1's profit is:

$$\pi_1 = [p'_1 + r_L - c_1] x_0 \le 0, \text{ where } x_0 \text{ is given by (10).}$$
(65)

The inequality in (65) holds because $p'_1 \leq c_1 - r_L$ and because (10) implies that $x_0 > 0$ in

the present setting. (65) implies that Firm 1 cannot secure strictly positive profit by setting $p'_1 \in (p_2 + G_1 - G_2 - K_1, p_2 + G_1 - G_2 + K_2)$ under the maintained conditions.

Next we show that Firm 1 cannot increase its profit by setting a price $p'_1 \leq p_2 + G_1 - G_2 - K_1$. Observe that under the maintained conditions:

$$c_1 - r_1 + K_1 + K_2 \ge p_1 > p_2 + G_1 - G_2 + K_2$$
(66)

$$\Rightarrow c_1 - r_1 > p_2 + G_1 - G_2 - K_1 \tag{67}$$

The weak inequality in (66) holds because $p_1 \in [c_1 - r_1, \min\{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$. The strict inequality in (66) holds because p_2 is marginally below $p_1 + G_2 - G_1 - K_2$. (67) implies that $p'_1 < c_1 - r_1$ if Firm 1 sets $p'_1 \leq p_2 + G_1 - G_2 - K_1$. Lemma A14 implies that this is a dominated strategy for Firm 1.

Now we show that Firm 1 cannot increase its profit by setting a price $p'_1 > p_2 + G_1 - G_2 + K_2$. Lemmas A7 and A8 imply that no user will purchase a phone from Firm 1 in this case. Consequently, Firm 1's profit is 0.

It remains to show that Firm 1 cannot increase its profit by setting a price $p'_1 = p_2 + G_1 - G_2 + K_2$. Because Firm 2 sets p_2 marginally below the p_2 in (62):

$$p'_1 \approx p_1 + G_2 - G_1 - K_2 + G_1 - G_2 + K_2 = p_1.$$
 (68)

Lemmas A1, A2, and A5 imply that when Firm 1 sets $p'_1 = p_2 + G_1 - G_2 + K_2$, all users located in $[0, \frac{1}{2} - \frac{K_2}{2t}]$ are indifferent between buying a phone from Firm 1 and from Firm 2, whereas all users located in $(\frac{1}{2} - \frac{K_2}{2t}, 1]$ purchase a phone from Firm 2. Consequently, (68) implies that Firm 1's profit in this case is:

$$\pi_1 = \frac{1}{2} \left[p_1' + r_L - c_1 \right] \left[\frac{1}{2} - \frac{K_2}{2t} \right] \approx \frac{1}{2} \left[p_1 + r_L - c_1 \right] \left[\frac{1}{2} - \frac{K_2}{2t} \right] \le 0.$$
 (69)

The inequality in (69) holds because $p_1 \leq c_1 - r_L$.

In summary, we have shown that if Firm 2 sets p_2 marginally below $p_1 + G_2 - G_1 - K_2$, then Firm 1 will set $p_1 \in [c_1 - r_1, min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$ in equilibrium.

We now show that Firm 2 maximizes its profit by setting p_2 marginally below $p_1 + G_2 - G_1 - K_2$ when Firm 1 sets $p_1 \in [c_1 - r_1, min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$. Observe first that this value of p_2 is positive because:

$$p_1 + G_2 - G_1 - K_2 \ge c_1 - r_1 + G_2 - G_1 - K_2 > 0.$$

The inequality here reflects Condition 3B.

When Firm 2 sets p_2 marginally below $p_1 + G_2 - G_1 - K_2$, all users purchase a phone from Firm 2. Lemma A1 implies that Firm 2's profit is:

$$\pi_{2} = \left[p_{2} + r_{H} - c_{2}\right] \left[\frac{1}{2} + \frac{K_{2}}{2t}\right] + \left[p_{2} + r_{L} - c_{2}\right] \left[\frac{1}{2} - \frac{K_{2}}{2t}\right]$$
$$\approx p_{1} + G_{2} - G_{1} - K_{2} - c_{2} + \frac{1}{2}\left[r_{L} + r_{H}\right] + \frac{K_{2}}{2t}\left[r_{H} - r_{L}\right].$$
(70)

In equilibria in which $p_1 \in [c_1 - r_1, \min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$ and Firm 2 sets p_2 is marginally below $p_1 + G_2 - G_1 - K_2$, Firm 2 secures the least profit when $p_1 = c_1 - r_1$. Consequently, (4) and (70) imply that Firm 2 earns positive profit in all such equilibria if:

$$\pi_2^{min} = G_2 - G_1 + c_1 - c_2 + \frac{K_1}{2t} \left[r_H - r_L \right] - K_2 \left[\frac{2t - (r_H - r_L)}{2t} \right].$$
(71)

(4) and Condition 3C imply that when $p_1 = c_1 - r_1$:

$$x_2 = \frac{1}{8t} \left[t + G_2 - G_1 + c_1 - c_2 + \left(\frac{1}{2} + \frac{K_1}{2t}\right) (r_H - r_L) \right]^2.$$
(72)

(4) and Condition 3C imply that when $p_1 = c_1 - r_1$:

$$\Omega_2(\cdot) = G_2 - G_1 + c_1 - c_2 + \frac{K_1}{2t} [r_H - r_L] - x_2.$$
(73)

(71) - (73) imply that when $p_1 = c_1 - r_1$ and Condition 3C holds:

$$\pi_2^{\min} > x_2 \ge 0.$$
 (74)

We now show that Firm 2 cannot increase its profit by setting $p_2 \in (p_1 + G_2 - G_1 - K_2, p_1 + G_2 - G_1 + K_1)$ or $p_2 \geq p_1 + G_2 - G_1 + K_1$ when Firm 1 sets $p_1 \in [c_1 - r_1, min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}].$

(35) and Lemma A11 imply that when $p_2 \in (p_1 + G_2 - G_1 - K_2, p_1 + G_2 - G_1 + K_1)$, the price that maximizes Firm 2's profit is:

$$p_2 = \frac{1}{2} \left[t + c_2 - r_H + G_2 - G_1 + p_1 \right].$$
(75)

(34) and (75) imply that Firm 2's corresponding profit is:

$$\pi_2' = \frac{1}{8t} \left[t + G_2 - G_1 - c_2 + r_H + p_1 \right]^2.$$
(76)

(74) and (76) imply that Firm 2 cannot increase its profit by setting $p_2 \in (p_1 + G_2 - G_1 - K_2 p_1 + G_2 - G_1 + K_1)$ when Condition 3C holds because:

$$\pi_2 \geq \pi'_2 \iff p_1 + G_2 - G_1 - K_2 - c_2 + \frac{1}{2} [r_L + r_H] + \frac{K_2}{2t} [r_H - r_L]$$

$$\geq \frac{1}{8t} [t + G_2 - G_1 - c_2 + r_H + p_1]^2 \iff K_2 \left[\frac{2t - r_H + r_L}{2t} \right] \leq \Omega_2.$$

Lemmas A9 and A10 imply that if Firm 2 sets $p_2 > p_1 + G_2 - G_1 + K_1$, it will sell no phones and so will secure 0 profit. Therefore, $p_2 = p_1 + G_2 - G_1 + K_1$ is the profitmaximizing price for Firm 2 among all $p_2 \ge p_1 + G_2 - G_1 + K_1$. Lemma A6 implies that when $p_2 = p_1 + G_2 - G_1 + K_1$: (i) all users located in $\left[\frac{1}{2} + \frac{K_1}{2t}, 1\right]$ are indifferent between buying a phone from Firm 1 and from Firm 2; and (ii) all users located in $\left[0, \frac{1}{2} + \frac{K_1}{2t}\right]$ buy a phone from Firm 1. Therefore, Lemma A1 implies that Firm 2's profit is:

$$\pi_2'' = \frac{1}{2} \left[p_1 + G_2 - G_1 + K_1 + r_H - c_2 \right] \left[\frac{1}{2} - \frac{K_1}{2t} \right] > 0.$$
(77)

The inequality in (77) holds because (4) implies that the minimum value of π_2'' , which occurs when $p_1 = c_1 - r_1$, is:

$$\pi_2^{''min} = \frac{1}{2} \left[G_2 - G_1 + c_1 - c_2 + K_1 + \left(\frac{1}{2} + \frac{K_1}{2t}\right) (r_H - r_L) \right] \left[\frac{1}{2} - \frac{K_1}{2t} \right] > 0.$$
 (78)

The inequality in (78) holds because: (i) $\frac{K_1}{2t} < \frac{1}{2}$, by assumption; and (ii) the term in the first square brackets in (78) is positive. (ii) holds because Condition 3B ensures this term exceeds: $\begin{bmatrix} 1 & K_1 \end{bmatrix}$

$$K_2 + K_1 + \left\lfloor \frac{1}{2} + \frac{K_1}{2t} \right\rfloor [r_H - r_L] > 0.$$

(78) ensures that (77) holds.

If Firm 2 were to reduce its price to $p_2 = p_1 + G_2 - G_1 + K_1 - \varepsilon_3$ where $\varepsilon_3 > 0$, all users located in $\left[\frac{1}{2} + \frac{K_1}{2t}, 1\right]$ would purchase a phone from Firm 2. Consequently, (77) implies that Firm 2's profit would be at least:

$$\pi_{2} = \left[p_{2} - \varepsilon_{3} + r_{H} - c_{2}\right] \left[\frac{1}{2} - \frac{K_{1}}{2t}\right]$$

$$= \left[p_{1} + G_{2} - G_{1} + K_{1} + r_{H} - c_{2}\right] \left[\frac{1}{2} - \frac{K_{1}}{2t}\right] - \varepsilon_{3} \left[\frac{1}{2} - \frac{K_{1}}{2t}\right]$$

$$= \pi_{2}'' + \pi_{2}'' - \varepsilon_{3} \left[\frac{1}{2} - \frac{K_{1}}{2t}\right] > \pi_{2}'' \text{ for sufficiently small } \varepsilon_{3}.$$
(79)

(79) implies that Firm 2 could increase its profit by reducing its price marginally below $p_1 + G_2 - G_1 + K_1$. Therefore, Firm 2 will never set $p_2 \ge p_1 + G_2 - G_1 + K_1$ when $p_1 \in [c_1 - r_1, \min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}].$

In summary, we have shown that when $p_1 \in [c_1 - r_1, min \{c_1 - r_L, c_1 - r_1 + K_1 + K_2\}]$, Firm 2 maximizes its profit by setting p_2 marginally below $p_1 + G_2 - G_1 - K_2$.

Finally, Lemma A1 implies that all users located in the interval $\left[0, \frac{1}{2} - \frac{K_2}{2t}\right]$ (and only these users) change the default setting on the phone they purchase from Firm 2.

Proof of Proposition 4. Conclusion (i) follows from Proposition 2, which implies that:

$$\pi_{1}^{D1}(\cdot) \approx G_{1} - G_{2} + c_{2} - c_{1} - \frac{1}{2} [r_{H} - r_{L}] - K_{1} \left[\frac{2t + r_{H} - r_{L}}{2t} \right]$$

$$\Rightarrow \frac{\partial \pi_{1}^{D1}(\cdot)}{\partial K_{1}} = -1 - \left[\frac{r_{H} - r_{L}}{2t} \right] < 0 \text{ and } \frac{\partial \pi_{1}^{D1}(\cdot)}{\partial K_{2}} = 0.$$
(80)

Conclusion (ii) follows from Proposition 3, which states that:

$$\underline{\pi}_{2}^{D2}(\cdot) \approx c_{1} - r_{1} - c_{2} + G_{2} - G_{1} - K_{2} + \frac{1}{2} [r_{L} + r_{H}] + \frac{K_{2}}{2t} [r_{H} - r_{L}]$$

$$\Rightarrow \frac{\partial \pi_{2}^{D2}(\cdot)}{\partial K_{2}} = -1 + \frac{r_{H} - r_{L}}{2t} \stackrel{\geq}{\geq} 0 \quad \Leftrightarrow \quad r_{H} - r_{L} \stackrel{\geq}{\geq} 2t; \text{ and} \qquad (81)$$

$$\frac{\partial -P^{2}(\cdot)}{\partial K_{2}} = -1 + \frac{1}{2t} \stackrel{\geq}{\leq} 0 \quad \Leftrightarrow \quad r_{H} - r_{L} \stackrel{\geq}{\leq} 2t; \text{ and} \qquad (81)$$

$$\frac{\partial \underline{\pi}_{2}^{D2}(\cdot)}{\partial K_{1}} = -\frac{\partial r_{1}}{\partial K_{1}} = \frac{1}{2t} [r_{H} - r_{L}] > 0.$$

$$(82)$$

Conclusion (iii) follows directly from Lemma 1 and Proposition 1.

<u>Proof of Lemma 3</u>. Observe that $A < 0 \Rightarrow G_1 - c_1 - (G_2 - c_2) > r_H - r_L > 0$. Therefore, Lemma 2 implies:

$$\pi_1^{D1}(0,0) = G_1 - c_1 - (G_2 - c_2) = -3A + \Delta.$$
(83)

(83) and Proposition 1 imply:

$$\pi_{1}^{S} \stackrel{\geq}{\geq} \pi_{1}^{D1}(0,0) \iff \frac{1}{2t} [t+|A|]^{2} \stackrel{\geq}{\geq} 3|A| + \Delta$$

$$\Leftrightarrow t^{2} - 2 [\Delta + 2|A|]t + |A|^{2} \stackrel{\geq}{\geq} 0.$$
(84)

(5) implies that the ("t") roots of the quadratic equation in (84) are given by:

$$t = \Delta + 2 |A| \pm \sqrt{3 |A|^2 + 4 |A| \Delta + \Delta^2} \in \{t_{1L}, t_{1H}\}.$$
 (85)

 t_{1L} , the smaller root in (85), satisfies the maintained assumption that t > |A| if and only if:

$$t_{1L} > |A| \iff \Delta + 2|A| - \sqrt{3|A|^2 + 4|A|} \Delta + \Delta^2 > |A|$$

$$\Leftrightarrow \Delta^2 + 2|A| \Delta + |A|^2 > 3|A|^2 + 4|A| \Delta + \Delta^2 \iff 2|A|^2 + 2|A| \Delta < 0.$$
(86)

This inequality does not hold.

(5) implies that t_{1H} satisfies the maintained assumption that t > |A| because:

$$t_{1H} > |A| \iff \Delta + |A| + \sqrt{3|A|^2 + 4|A|\Delta + \Delta^2} > 0.$$
 (87)

(84) – (87) imply that because t > |A| by assumption, $\pi_1^S - \pi_1^{D1}(0,0)$ is: (i) negative for $t < t_{1H}$; and (ii) positive for $t > t_{1H}$. Formally:

$$\pi_1^S > \pi_1^{D1}(0,0)$$
 if $t > t_{1H}$ and $\pi_1^S < \pi_1^{D1}(0,0)$ if $t \in (K_1, t_{1H})$. (88)

The lower bound on t in the second conclusion in (88) (i.e., K_1) reflects the maintained assumption that $K_1 < t$.

(83) and Proposition 1 imply:

$$\pi_1^S - \pi_1^{D1}(0,0) = \frac{1}{2t} \left[t + |A| \right]^2 - 3|A| - \Delta.$$
(89)

(89) implies:

$$\frac{\partial \left(\pi_{1}^{S} - \pi_{1}^{D1}(0,0)\right)}{\partial |A|} = \frac{1}{t} \left[t + |A|\right] - 3 = \frac{|A|}{t} - 2 \leq 0 \iff |A| < 2t$$
$$\Rightarrow \frac{\partial \left(\pi_{1}^{S} - \pi_{1}^{D1}(0,0)\right)}{\partial |A|} < 0 \text{ for all } |A| < t.$$
(90)

The expression in (84) can be written as:

$$|A|^2 - 4t |A| + t^2 - 2\Delta t \stackrel{\geq}{<} 0.$$

(5) implies that the ("|A|") roots of the associated quadratic equation are:

$$|A| = \frac{1}{2} \left[4t \pm \sqrt{16t^2 - 4t[t - 2\Delta]} \right] = 2t \pm \sqrt{3t^2 + 2t\Delta} \in \{ |A|_{1L}, |A|_{1H} \}.$$
(91)

(5) implies:

$$|A|_{1L} > 0 \iff 2t > \sqrt{3t^2 + 2t\Delta} \iff t^2 > 2t\Delta \iff t > 2\Delta.$$
(92)

(5) further implies that $|A|_{1L} < \frac{t}{2}$ because:

$$|A|_{1L} < \frac{t}{2} \Leftrightarrow 2t - \sqrt{3t^2 + 2t\Delta} < \frac{t}{2} \Leftrightarrow \frac{3}{4}t^2 + 2t\Delta > 0.$$
 (93)

(5) also implies that $|A|_{1H} > t$ because:

$$|A|_{1H} > t \Leftrightarrow 2t + \sqrt{3t^2 + 2t\Delta} > t \Leftrightarrow t + \sqrt{3t^2 + 2t\Delta} > 0.$$
(94)

(84) and (90) – (94) imply that $\pi_1^S - \pi_1^{D1}(0,0)$ is a decreasing function of |A| for |A| < t. Furthermore, this function is: (i) positive when $|A| < |A|_{1L}$ (which can occur if and only if $t > 2\Delta$); and (ii) negative for $|A| \in (|A|_{1L}, t)$ (because $t > |A|_{1H}$, by assumption). Formally:

If
$$t > 2\Delta$$
, then $\pi_1^S \begin{cases} > \pi_1^{D1}(0,0) \text{ if } |A| < |A|_{1L} \\ < \pi_1^{D1}(0,0) \text{ if } |A| \in (|A|_{1L}, t). \end{cases}$
If $t \le 2\Delta$, then $\pi_1^S < \pi_1^{D1}(0,0)$ for all $|A| \in (0, t).$ (95)

The proof of Lemma 4 parallels the proof of Lemma 3, and so is omitted.

Proof of Proposition 5. Condition (i) in the proposition implies:

$$A \equiv \frac{1}{3} \left[G_2 + r_H - c_2 - (G_1 + r_L - c_1) \right] = \frac{1}{3} \left[G_2 - c_2 - (G_1 - c_1) \right] + \frac{1}{3} \left[r_H - r_L \right] > 0.$$

Therefore, $K_1 > -A$ when $K_1 = 0$. Furthermore, condition (ii) ensures that t > |A| = Aand $r_H - c_2 - t < A < t + c_1 - r_L$. Therefore, all the maintained assumptions in Proposition 1 hold if $K_2 \in (A, t)$.

Lemma A2.3 implies that when $K_1 = K_2 = 0$, Firm 2's profit is nearly:

$$\pi_{2A} = G_2 - c_2 - (G_1 - c_1) = 3A - (r_H - r_L).$$
(96)

Proposition 1 implies that when $K_2 \in (A, t)$, there exists an equilibrium in which Firm 2's profit is:

$$\pi_{2B} = \frac{[t+A]^2}{2t}.$$
(97)

(96) and (97) imply:

$$\pi_{2B} > \pi_{2A} \iff t^2 + 2t [r_H - r_L - 2A] + A^2 > 0.$$
(98)

The roots of the quadratic equation associated with (98) are:

$$t^* = 2A - \Delta \pm \sqrt{[3A - \Delta][A - \Delta]}.$$
⁽⁹⁹⁾

Condition (ii) ensures that $A > \Delta$, which implies $3A > \Delta$, so the roots in (99) are real. (5), (98), and (99) imply that $\pi_{2B} > \pi_{2A}$ if $t > 2A - \Delta + \sqrt{[3A - \Delta][A - \Delta]} = t_{2H}$.

Finally, Lemma A2.3 implies that Firm 1's equilibrium profit is 0 when $K_1 = K_2 = 0$. Firm 1's profit in the equilibrium identified in Proposition 1 is $\frac{1}{18t} [3t - A]^2 > 0$ when $K_1 = 0$ and $K_2 \in (A, t)$.

Proof of Proposition 6. Condition (i) implies that $K_2 > A$ when $K_2 = 0$. Therefore, conditions (i) – (iii) ensure that all the conditions in Proposition 1 hold if $K_1 \in (|A|, t)$.

Lemma 2 implies that when $K_1 = K_2 = 0$, Firm 1's profit is nearly:

$$\pi_{1A} = G_1 - c_1 - (G_2 - c_2) = r_H - r_L + 3 |A|.$$
(100)

Proposition 1 implies that when $K_1 \in (|A|, t)$, there exists an equilibrium in which Firm 1's profit is:

$$\pi_{1B} = \frac{[t+|A|]^2}{2t}.$$
(101)

(100) and (101) imply:

$$\pi_{1B} > \pi_{1A} \iff t^2 - 2t [r_H - r_L + 2|A|] + A^2 > 0.$$
(102)

The roots of the quadratic equation associated with (102) are:

$$t_1^* = \Delta + 2 |A| \pm \sqrt{3A^2 + 4 |A| \Delta + \Delta^2}.$$
 (103)

(5), (102), and (103) imply that if $t > \Delta + 2 |A| \pm \sqrt{3A^2 + 4 |A| \Delta + \Delta^2} = t_{1H}$, then $\pi_{1B} > \pi_{1A}$.

Finally, Lemma 2 implies that Firm 2's equilibrium profit is 0 when $K_1 = K_2 = 0$. Firm 2's profit in the equilibrium identified in Proposition 1 is $\frac{1}{18t} [3t - |A|]^2 > 0$ when $K_2 = 0$ and $K_1 \in (|A|, t)$.

Proof of Proposition 7. The proof proceeds by showing that neither firm can increase its profit by unilaterally changing its default-switching cost, regardless of the nature of the ensuing equilibrium.

To begin, observe that the proof of Proposition 1 establishes that the prices identified in the present Proposition are the unique prices that arise in a MS equilibrium. Furthermore, the equilibrium profits identified in Proposition 1 ($\pi_1^S > 0$ and $\pi_2^S > 0$) do not vary with K_1 and K_2 . Therefore, Firm $i \in \{1, 2\}$ cannot increase its profit by choosing $K_i \neq K_i^*$ if the resulting (K_i, K_i^*) default-switching costs induce a MS equilibrium.

Next suppose that Firm $i \in \{1, 2\}$ chooses a $K_i \neq K_i^*$ such that the resulting (K_i, K_j^*) default-switching costs induce a MD*j* equilibrium (where $j \neq i, i, j \in \{1, 2\}$). Then Firm *i*'s profit will decline to 0. Consequently, Firm *i* cannot increase its profit by setting $K_i \neq K_i^*$ if the resulting (K_i, K_j^*) default-switching costs induce a MD*j* equilibrium.

Now suppose that Firm 1 sets $K_1 \neq K_1^*$ such that the resulting (K_1, K_2^*) default-switching costs induce a MD1 equilibrium. The proof of Proposition 1 establishes that the maximum profit Firm 1 can secure in a MD1 equilibrium is

$$-2A + \frac{t - K_1}{2t} \left[2t + \Delta \right] \leq -2A + \frac{1}{2} \left[2t + \Delta \right].$$

Condition (ii) in the present Proposition ensures that Firm 1's profit in the MS equilibrium identified in Proposition 1 exceeds $-2A + \frac{1}{2}[2t + \Delta]$. Therefore, Firm 1 cannot increase its profit by setting $K_1 \neq K_1^*$ if the resulting (K_1, K_2^*) default-switching costs induce a MD1 equilibrium.

Finally, suppose that Firm 2 sets a $K_2 \neq K_2^*$ such that the resulting (K_1^*, K_2) defaultswitching costs induce a MD2 equilibrium. The proof of Proposition 1 establishes that the maximum profit Firm 2 can secure in a MD2 equilibrium is

$$2A + \frac{t - K_2}{2t} \left[2t - \Delta \right] \leq 2A + \frac{1}{2} \left[2t - \Delta \right].$$

The inequality here reflects condition (iii) in the present Proposition. Condition (ii) in this Proposition ensures that Firm 2's profit in the MS equilibrium identified in Proposition 1 exceeds $2A + \frac{1}{2} [2t - \Delta]$. Therefore, Firm 2 cannot increase its profit by setting $K_2 \neq K_2^*$ if the resulting (K_1^*, K_2) default-switching costs induce a MD2 equilibrium.

Proof of Proposition 8. The proof proceeds by showing that neither firm can increase its profit by unilaterally increasing its default-switching cost above 0, regardless of the nature of the ensuing equilibrium.

We first show that Firm 1 cannot strictly increase its profit by setting $K_1 > 0$ if the resulting $(K_1, 0)$ default-switching costs induce a MD1 equilibrium. The logic employed in the proof of Proposition 2 implies that Firm 1's profit in a MD1 equilibrium, given $K_1 > 0$ and $K_2 \ge 0$ is:

$$\pi_1^{D1}(K_1, K_2) = G_1 - G_2 + c_2 - c_1 - \frac{\Delta}{2} - K_1 \left[\frac{2t + \Delta}{2t} \right] \quad \Rightarrow \quad \frac{\partial \pi_1^{D1}(\cdot)}{\partial K_1} < 0.$$
(104)

(104) implies that Firm 1 cannot strictly increase its profit setting $K_1 > 0$ if a MD1 equilibrium ensues.

(104) further implies that when $K_1 = K_2 = 0$, Firm 1's profit is:

$$\pi_1^{D1}(0,0) = G_1 - c_1 - (G_2 - c_2) - \frac{\Delta}{2} > 0.$$
 (105)

Firm 1's profit is 0 in any MD2 equilibrium. Therefore, (105) implies that Firm 1 cannot increase its profit by setting $K_1 > 0$ if the resulting $(K_1, 0)$ default-switching costs induce a MD2 equilibrium.

Proposition 1 establishes that Firm 1's profit in a MS equilibrium when A < 0 is $\pi_1^S = \frac{1}{2t} [t + |A|]^2$. Therefore, (105) and the maintained assumptions ensure that $\pi_1^{D1}(0,0) > \pi_1^S$. Consequently, Firm 1 cannot increase its profit by setting $K_1 > 0$ if the resulting $(K_1,0)$ default-switching costs induce a MS equilibrium.

To initiate the demonstration that Firm 2 cannot increase its profit by unilaterally increasing K_2 , observe that Firm 2's profit is 0 in all MD1 equilibria. Consequently, Firm 2 cannot increase its profit by implementing a $K_2 > 0$ that induces a MD1 equilibrium.

Next we establish that a MD2 equilibrium does not exist when $K_1 = 0$ and A < 0. To do so, suppose such an equilibrium exists. Then the consumer located at 0 prefers to buy a phone from Firm 2 than from Firm 1. Consequently:

$$G_2 - p_2 - \min\{K_2, t\} \ge G_1 - p_1 \implies p_2 \le p_1 + G_2 - G_1 - \min\{K_2, t\}.$$
 (106)

(106) reflects the fact that the consumer located at 0 who purchases a phone from Firm 2 will change the default PD setting on the phone if and only if $K_2 < t$.

Rather than serve no customers, Firm 1 will reduce its price at least to $c_1 - r_L$. Therefore, (106) implies that, to attract all consumers, Firm 2's price must satisfy:

$$p_2 \leq c_1 - r_L + G_2 - G_1 - \min\{K_2, t\}.$$
(107)

In any MD2 equilibrium in which (107) holds, Firm 2's profit is:

$$\pi_2 \leq c_1 - r_L + G_2 - G_1 - \min\{K_2, t\} - c_2 + r_H$$

$$< c_1 - r_L + G_2 - G_1 - c_2 + r_H = 3A < 0.$$
(108)

The first inequality in (108) reflects the fact that Firm 2's revenue from advertisers cannot exceed r_H . The second inequality in (108) holds because $K_2 > 0$ (and t > 0), by assumption. The last inequality in (108) holds because A < 0, by assumption. (108) implies that a MD2 equilibrium does not exist under the maintained assumptions because Firm 2 must secure nonnegative profit in a MD2 equilibrium.

Finally, we establish that a MS equilibrium does not exist when $K_1 = 0$ and A < 0. To do so, suppose a MS equilibrium exists. Then there exists a consumer located at $x_0 \in (0, 1)$ who is indifferent between purchasing a phone from Firm 1 and purchasing a phone from Firm 2. Furthermore, the proof of Proposition 1 implies:

$$p_1 = c_1 - r_L + t - A, \quad p_2 = c_2 - r_H + t + A, \text{ and}$$
(109)

$$x_0 = \frac{1}{2} - \frac{A}{2t} > \frac{1}{2}.$$
(110)

The inequality in (110) holds because A < 0, by assumption.

Because $x_0 > \frac{1}{2}$ and $K_1 = 0$, the consumer located at x_0 will change the default PD setting on the phone he purchases if and only if he buys the phone from Firm 1. Therefore, because the consumer located at x_0 is indifferent between purchasing a phone from Firm 1 and purchasing a phone from Firm 2:

$$G_1 - p_1 - t [1 - x_0] = G_2 - p_2 - t [1 - x_0] \implies p_2 - p_1 = G_2 - G_1.$$
(111)

(109) implies:

$$p_2 - p_1 = c_2 - c_1 - r_H + r_L + 2A.$$
(112)

(111) and (112) imply:

 $G_2 - G_1 = c_2 - c_1 - r_H + r_L + 2A \implies 3A = 2A \implies A = 0.$ (113)

(113) cannot hold because A < 0, by assumption. Therefore, by contradiction, a MS equi-

librium does not exist when $K_1 = 0$ and A < 0.

The proofs of Propositions 9 and 10 parallel the proof of Proposition 8, and so are omitted.

Proof of Proposition 11. First consider a putative MD1 equilibrium in which $K_1 \in (0, t)$ and $K_2 \ge 0$. Arguments analogous to those employed in the proof of Proposition 2 reveal that Firm 1 can increase its profit by reducing K_1 marginally. Therefore, the putative equilibrium cannot constitute an equilibrium.

Next consider a putative MD1 equilibrium in which $K_1 \ge t$ and $K_2 \ge 0$. It is readily verified that Firm 1's profit in this equilibrium is $\pi_1^{D1} = -3A - t$. Proposition 2 implies that if Firm 1 reduces K_1 below t, it can secure a profit of nearly $\pi_1^{D1'} = \frac{t-K_1}{2t} \Delta - 3A - K_1$. Observe that when $K_1 \in (0, t)$:

$$\pi_1^{D1'} > \pi_1^{D1} \Leftrightarrow \frac{t - K_1}{2t} \Delta - 3A - K_1 > -3A - t \Leftrightarrow \frac{\Delta}{2t} + 1 > 0.$$
(114)

Because the last inequality in (114) always holds, (114) implies that the putative equilibrium cannot constitute an equilibrium.

Proof of Proposition 12. First consider a putative MD2 equilibrium in which $K_1 \in [0, t)$ and $K_2 \in (0, t)$, $p_1 = c_1 - r_1$, and p_2 is marginally below $c_1 - r_1 + G_2 - G_1 - K_2$. The expression for π_2 in Proposition 3 implies that Firm 2 can increase its MD2 equilibrium profit by reducing K_2 marginally. Consequently, the identified putative equilibrium cannot constitute an equilibrium under the maintained conditions.

Now consider a putative MD2 equilibrium in which $K_1 \in [0, t), K_2 \geq t, p_1 = c_1 - r_1$, and p_2 is marginally below $c_1 - r_1 + G_2 - G_1 - K_2$. Arguments analogous to those employed in the proof of Proposition 3 reveal that Firm 2's profit in this equilibrium is $\pi_2^{D2} = c_1 - r_1 + r_H - c_2 + G_2 - G_1 - t$. Proposition 3 implies that if Firm 2 reduces K_2 below t, its profit is nearly $\pi_2^{D2'} = c_1 - r_1 - c_2 + G_2 - G_1 - K_2 + \frac{1}{2} [r_H + r_L] + \frac{K_2}{2t} \Delta$. Observe that that when $K_2 \in (0, t)$:

$$\pi_2^{D2} < \pi_2^{D2'} \Leftrightarrow r_H - t < -K_2 + \frac{1}{2} [r_H + r_L] + \frac{K_2}{2t} \Delta \Leftrightarrow 2t > \Delta.$$
(115)

The last inequality in (115) holds, by assumption. Therefore, (115) implies that the identified putative equilibrium cannot constitute an equilibrium under the maintained conditions.

Proof of Proposition 13. First consider a putative equilibrium in which: (i) $K_1 \in [0, t)$ and $K_2 \in [0, t)$, where $(K_1, K_2) \neq (0, 0)$; (ii) $p_1 = c_1 - r_1$; and (iii) p_2 is marginally below $c_1 - r_1 + G_2 - G_1 - K_2$. The expression for π_2 in Proposition 3 implies that Firm 2 can increase its MD2 equilibrium profit by reducing K_2 marginally. Therefore, the identified putative equilibrium cannot constitute an equilibrium.

Now consider a putative equilibrium in which $K_1 = K_2 = 0$, $p_1 = c_1 - r_1$, and $p_2 = c_1 - \frac{1}{2} [r_L + r_H] + G_2 - G_1$. Arguments analogous to those employed in the proof of Proposition 3 reveal that Firm 2's profit in this equilibrium is nearly $\pi_2^{D2} = G_2 - c_2 - (G_1 - c_1)$. Proposition 3 implies that if Firm 2 increases K_2 marginally and reduces its price to $p_2 = c_1 - \frac{1}{2} [r_L + r_H] + G_2 - G_1 - \varepsilon_4$ (where $\varepsilon_4 > 0$ is arbitrarily small), then its profit would be nearly $\pi_2^{D2'} = G_2 - G_1 + c_1 - c_2 - \varepsilon_4 + \frac{\varepsilon_4}{2t} \Delta$. Observe that:

$$\pi_2^{D2} < \pi_2^{D2'} \Leftrightarrow -\varepsilon_4 + \frac{\varepsilon_4}{2t} \Delta > 0 \Leftrightarrow \frac{\Delta}{2t} > 1 \Leftrightarrow \Delta > 2t.$$
(116)

The last inequality in (116) holds, by assumption. Therefore, (116) implies that the identified putative equilibrium cannot constitute an equilibrium under the maintained conditions.

References

Acquisti, Alessandro and Jens Grossklags, "Privacy and Rationality in Individual Decision Making," *IEEE Security & Privacy*, 3(1), January-February 2005, 26-33.

Acquisti, Alessandro, Curtis Taylor, and Liad Wagman, "The Economics of Privacy," *Journal of Economic Literature*, 54(2), June 2016, 442-492.

Adlucent, "71% of Consumers Prefer Personalized Ads," visited February 4, 2024 (https://www.adlucent.com/resources/blog/71-of-consumers-prefer-personalized-ads/).

Arie, Guy and Paul Grieco, "Who Pays for Switching Costs?" Quantitative Marketing and Economics, 12, 2014, 379-419.

Athey, Susan, Christian Catalini, and Catherine Tucker, "The Digital Privacy Paradox: Small Money, Small Costs, Small Talk," NBER Working Paper Series No. 23488, June 2017 (https://www.nber.org/papers/w23488).

Arora, Nidhi, Daniel Ensslen, Lars Fiedler, Wei Wei Liu, Kelsey Robinson, Eli Stein, and Gustavo Schüler, "The Value of Getting Personalization Right – or Wrong – is Multiplying," McKinsey & Company Report, November 12, 2021 (https://www.mckinsey.com/capabilities/growth-marketing-and-sales/our-insights/the-value-of-getting-personalization-right-or-wrong -is-multiplying).

Australian Competition and Consumer Commission, "Digital Platform Services Inquiry: Interim Report No. 3 – Search Defaults and Choice Screens," September 2021 (https://www. accc.gov.au/inquiries-and-consultations/digital-platform-services-inquiry-2020-25/september-2021-interim-report).

Beales, Howard, "The Value of Behavioral Targeting," Supplementary Comments of the Network Advertising Initiative Before the Federal Trade Commission, Project No. P095416, April 8, 2010 (https://www.ftc.gov/sites/default/files/documents/public_comments/privacy-roundtables-comment-project-no.p095416-544506-00117/544506-00117.pdf).

Beyond Identity, "A Digital Dive into Brand Security (Survey)," Beyond Identity Blog, July 2022 (https://www.beyondidentity.com/resource/a-digital-dive-into-brand-security-survey# :~:text=Consumers%20reported%20feeling%20most%20secure,superiority%20in%20security %20and%20privacy). Accessed June 25, 2024.

Biscaia, Ricardo and Isabel Mota, "Models of Spatial Competition: A Critical Review," *Papers in Regional Science*, 92(4), November 2013, 851-871.

Casadesus-Masanell, Ramon and Andres Hervas-Drane, "Competing with Privacy," *Management Science*, 61(1), January 2015, 229-246.

Chakravorty, Shourjo and David Sappington, "Technical Appendix to Accompany 'Hotelling Competition with Avoidable Horizontal Product Differentiation: Choosing Between Online Privacy and Disclosure'," June 2024 (https://people.clas.ufl.edu/sapping). Chellappa, Ramnath and Raymond Sin, "Personalization versus Privacy: An Empirical Examination of the Online Consumer's Dilemma," *Information Technology and Management*, 6, 2005, 181-202.

Competition and Markets Authority, "Online Platforms and Digital Advertising: Market Study Final Report," July 1, 2020 (https://www.gov.uk/cma-cases/online-platforms-and-digital-advertising-market-study).

Competition and Markets Authority, "Consumer Purchasing Behaviour in the UK Smartphone Market for the CMA's Mobile Ecosystems Market Study: Final Report," June 2022 (https://www.gov.uk/find-digital-market-research/consumer-purchasing-behaviour-in-the-uksmartphone-market-cma-research-report-2022-cma).

Danao, Monique and Kelly Main, "PPC Advertising: The Ultimate Guide,' *Forbes*, last updated February 1, 2024 (https://www.forbes.com/advisor/business/ppc-marketing-guide/).

Farahat, Ayman and Michael Bailey, "How Effective is Targeted Advertising?" *Proceedings of the 21st International Conference on World Wide Web*, 2012, 111-120 (https://dl.acm.org/doi/proceedings/10.1145/2187836).

Gabszewicz, Jean and Jacques-François Thisse, "Location," in Robert Aumann and Sergiu Hart (eds.), *Handbook of Game Theory with Economic Applications*, Volume 1. Amsterdam: North-Holland, 1992, pp. 281-304.

Gal-Or, Ester, Ronen Gal-Or, and Nabita Penmetsa, "The Role of User Privacy Concerns in Shaping Competition Among Platforms," *Information Systems Research*, 29(3), 2018, 698-722.

Gehrig, Thomas and Rune Stenbacka, "Differentiation-induced Switching Costs and Poaching," *Journal of Economics and Management Strategy*, 13(4), 2004, 635-655.

Graitson, Dominique, "Spatial Competition à la Hotelling: A Selective Survey," Journal of Industrial Economics, 31(1/2), September-December 1982, 11-25.

Grant, Nico and *Bloomberg*, "Google to Let Android Users Opt Out of Tracking, Following Apple," *Fortune*, June 4, 2021 (https://fortune.com/2021/06/03/google-android-users-opt-out-of-tracking-apple).

Hendel, Igal and John Neiva de Figueiredo, "Product Differentiation and Endogenous Disutility," *International Journal of Industrial Organization*, 16(1), November 1997, 63-79.

Hou, Haiyang, Xiaobo Wu, and Weihua Zhou, "The Competition of Investments for Endogenous Transportation Costs in a Spatial Model," *Economic Modelling*, 31, March 2013, 574-577.

Hutton, Hannah and David Ellis, "Exploring User Motivations Behind iOS App Tracking Transparency Decisions," *Proceedings of the 2023 CHI Conference on Human Factors in Computing Systems*, 2023, 1-12.

Mueller, Benjamin and Daniel Castro, "The Value of Personalized Advertising in Europe," Center for Data Innovation Report, November 22, 2021 (https://www2.datainnovation.org/2021-value-personalized-ads-europe.pdf).

O'Flaherty, Kate, "iOS 15: Apple Gives iPhone Users Opt-in to its Own Ads," *Forbes*, September 4, 2021 (https://www.forbes.com/sites/kateoflahertyuk/2021/09/04/ios-15-apple-just-revealed-a-game-changing-new-iphone-privacy-feature/?sh=26f56c19115a).

Rhodes, Andrew, "Re-examining the Effect of Switching Costs," *Economic Theory*, 57(1), 2014, 161-194.

Rhodes, Andrew and Jidong Zhou, "Personalized Pricing and Privacy Choice," Toulouse School of Economics discussion paper, September 2022 (https://www.tse-fr.eu/sites/default/files/TSE/documents/doc/wp/2022/wp_tse_1333.pdf).

Salesforce Research, "Trends in Customer Trust," visited February 4, 2024 (https://www.sales force.com/form/conf/trust-research/).

Samsung, "Samsung Privacy Policy for the U.S.," last updated December 31, 2023 (https://www.samsung.com/us/account/privacy-policy/#:~:text=Our%20Services%20collect%20 some%20data,web%20beacons%2C%20and%20other%20technologies).

Statcounter, "Mobile Operating System Market Share United States of America: August 2022 – August 2023," *GlobalStats*, visited September 12, 2023 (https://gs.statcounter.com/os-market-share/mobile/united-states-of-america).

Statista, "Attitudes towards Personalized Ads in the US 2022," visited February 4, 2024 (https://www.statista.com/statistics/796167/us-internet-users-primary-attitude-personalize d-video-ads/).

Statista, "Smartphones in the U.S. – Statistics and Facts," *Telecommunications*, visited September 12, 2023 (https://www.statista.com/topics/2711/us-smartphone-market/#topicOverview).

Taylor, Curtis and Liad Wagman, "Consumer Privacy in Oligopolistic Markets: Winners, Losers, and Welfare," *International Journal of Industrial Organization*, 34, May 2014, 80-84.

Troncoso-Valverde, Cristian and Jacques Robert, "On Hotelling's Competition with General Purpose Products," FACE Working Paper Series No. 3, November 2004 (https://ssrn.com/abstract=719770 or http://dx.doi.org/10.2139/ssrn.719770).

Tucker, Catherine, "The Economics of Advertising and Privacy," *International Journal of Industrial Organization*, 30(3), May 2012, 326-329.

Turow, Joseph, Jennifer King, Chris Jay Hoofnagle, Amy Bleakley, and Michael Hennessy. "Americans Reject Tailored Advertising and Three Activities that Enable It," Available at SSRN 1478214, September 29, 2009 (https://papers.ssrn.com/sol3/papers.cfm?abstract_id= 1478214).

Von Ungern-Sternberg, Thomas, "Monopolistic Competition and General Purpose Products," *Review of Economic Studies*, 55(2), April 1988, 231-246.

Wilson, Chris, "Ordered Search and Equilibrium Obfuscation," International Journal of Industrial Organization, 28(5), September 2010, 496-506.

Yan, Jun, Ning Liu, Gang Wang, Wen Zhang, Yun Jiang, and Zheng Chen, "How Much Can Behavioural Targeting Help Online Advertising?" *Proceedings of the 18th International Conference on World Wide Web*, 2009, 261-270 (https://dl.acm.org/doi/proceedings/10.1145/152 6709).